

LUND UNIVERSITY

Advanced patterning and processing for III-V nanowire device fabrication

Jafari Jam, Reza

2019

Document Version: Publisher's PDF, also known as Version of record

Link to publication

Citation for published version (APA): Jafari Jam, R. (2019). Advanced patterning and processing for III-V nanowire device fabrication. [Doctoral Thesis (compilation), Faculty of Engineering, LTH]. Department of Physics, Lund University.

Total number of authors:

General rights

Unless other specific re-use rights are stated the following general rights apply:

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights. • Users may download and print one copy of any publication from the public portal for the purpose of private study

- or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal

Read more about Creative commons licenses: https://creativecommons.org/licenses/

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

LUND UNIVERSITY

PO Box 117 221 00 Lund +46 46-222 00 00

Advanced patterning and processing for III-V nanowire device fabrication

REZA JAFARI JAM DEPARTMENT OF PHYSICS | FACULTY OF ENGINEERING | LUND UNIVERSITY



Advanced patterning and processing for III-V nanowire device fabrication

Reza Jafari Jam



DOCTORAL DISSERTATION

by due permission of the Faculty of engineering at Lund University, Sweden. To be publicly defended on Thursday, January 30th, 2020 at 13:15 in Rydberg Hall at the Department of Physics, Sölvegatan 14, Lund, for the degree of Doctor of Philosophy in Engineering.

> *Faculty opponent:* Prof. Ritesh Agarwal University of Pennsylvania

Organization		Document name		
LUND UNIVERSITY		Doctoral thesis		
		Date of issue: 2019	9-12-20	
Reza Jafari Jam		Sponsoring organiz	zation	
Title and subtitle:				
Advanced patterning and proces	sing for	III-V nanowire device f	abrication	
Abstract				
Semiconductor nanowires are widely considered as promising candidates for next generations of electronics and optoelectronics. Gold seed particles have so far been recognized as the most important catalyst for growth of nanowires. Costs related to substrates, lithography processes and gold consumption most strongly influence the final cost of nanowire-based devices. An optimum gold deposition process can facilitate reuse of substrates, avoid repetitive lithography processes and reduce the gold consumption in order to reduce the cost of the final device. Moreover, a novel deposition process is needed to deposit gold seeds in complex nanoscale templates used for guided growth of nanowires, and subsequent device fabrication, on commercially viable (001)-oriented Si and SOI substrates.				
In this work, we report on nanowire seed definition by gold electrodeposition. Electron microscopy inspection and optical spectroscopy confirm that the subsequently grown III-V nanowires have surface morphology and crystal quality comparable to those of nanowires grown from seed particles defined by conventional thermal evaporation. For seeding of substrates used in a typical growth run, we used around 650 times less gold compared to thermal evaporation.				
We also demonstrated the possibility of depositing seed particles in deep holes etched in thick templates to facilitate template-assisted growth of nanowires on (001) Si and InP substrates. In addition to a functional p-n junction on an (001) InP substrate, the first reported InP-InAs-InP axial heterostructures on an (001) substrate was demonstrated. In order to reuse the substrate for multiple growth runs, growth of AlAs-GaAs nanowires for epitaxial lift-off was realized. Using selective gold electrodeposition, we demonstrated deposition of seed particles on a substrate after nanowire peel-off. AlAs-GaAs nanowires were subsequently regrown demonstrating the potential of this novel technique for substrate reuse.				
Key words		la stra dan saiti an O	Stick Line of the test of the test	
Nanowires, Template-assisted growth, Electrodeposition, Sacrifitial layer, Substrate reuse				
Classification system and/or index terms (if any)				
Supplementary bibliographical information		Language English		
ISSN and key title			ISBN 978-91-7895-404-9 (Print) ISBN 978-91-7895-405-6 (Electronic)	
Recipient's notes	Num	per of pages 67	Price	
	Secu	rity classification		
		-		

I, the undersigned, being the copyright owner of the abstract of the above-mentioned dissertation, hereby grant to all reference sources permission to publish and disseminate the abstract of the above-mentioned dissertation.

Øe,

Signature

Date 2019-12-09

Advanced patterning and processing for III-V nanowire device fabrication

Reza Jafari Jam



Cover photo: Cross-sectional SEM micrograph of a GaAs-AlAs-GaAs NW heterostructure which is used to facilitate substrate reuse. One of the NWs is artificially coloured to highlight the different materials along the NW.

Copyright pp 1-67 (Reza Jafari Jam) Paper 1 © 2015 American Chemical Society Paper 2 © 2019 The Royal Society of Chemistry Paper 3 © 2019 R. J. Jam et.al. (Manuscript unpublished) Paper 4 © 2019 V. J. Gómez et.al. (Manuscript unpublished)

Faculty of Engineering Department of Physics

ISBN 978-91-7895-404-9 (Print) ISBN 978-91-7895-405-6 (Electronic)

Printed in Sweden by Media-Tryck, Lund University Lund 2020



To the amazing women of my life, my mother and my wife

Content

	Abstrac	et				
	Acknow	wledgment	9			
	Popular science summary					
	List of Papers					
	Abbrev	viations				
Intro	oduction		17			
1	Gold e	lectrodeposition	19			
	1.1.	Introduction				
	1.2.	Direct current electrodeposition				
	1.3.	Choice of anode				
	1.4.	Electrodeposition baths				
	1.5.	Pulse electrodeposition				
	1.6.	Electrodeposition in the electronics industry				
2	Gold e	lectrodeposition for III-V nanowire seed definition				
	2.1.	Introduction				
	2.2.	Gold baths				
	2.3.	Sample preparation				
	2.4.	Deposition mechanism				
	2.5.	Electrodeposition on Si substrates				
	2.6.	Electrodeposition on III-V substrates				
	2.7.	Electrodeposition in high aspect ratio features				
	2.8.	Large scale electrodeposition	40			

3	Nano	Nanowire synthesis	
	3.1.	Introduction	41
	3.2.	Vapor-Liquid-Solid growth mode	41
	3.3.	Nanowire growth by use of electrodeposited Au particles	42
	3.4.	Template-assisted VLS growth on (001) substrates	46
4	Subst	rate reuse	53
	4.1.	Introduction	53
	4.2.	Growth of GaAs/AlAs nanowires on GaAs substrates	54
	4.3.	Nanowire peel-off	56
	4.4.	Etching of sacrificial AlAs segment	57
	4.5.	Gold electrodeposition	58
	4.6.	Regrowth of AlAs/GaAs nanowires	58
5	Conc	lusions and Outlook	61
6	Refer	ences	63

Abstract

Semiconductor nanowires are widely considered as promising candidates for next generations of electronics and optoelectronics. Gold seed particles have so far been recognized as the most important catalyst for growth of nanowires. Costs related to substrates, lithography processes and gold consumption most strongly influence the final cost of nanowire-based devices. An optimum gold deposition process can facilitate reuse of substrates, avoid repetitive lithography processes and reduce the gold consumption in order to reduce the cost of the final device. Moreover, a novel deposition process is needed to deposit gold seeds in complex nanoscale templates used for guided growth of nanowiress, and subsequent device fabrication, on commercially viable (001)-oriented Si and SOI substrates.

In this work, we report on nanowire seed definition by gold electrodeposition. Electron microscopy inspection and optical spectroscopy confirm that the subsequently grown III-V nanowires have surface morphology and crystal quality comparable to those of nanowires grown from seed particles defined by conventional thermal evaporation. For seeding of substrates used in a typical growth run, we used around 650 times less gold compared to thermal evaporation.

We also demonstrated the possibility of depositing seed particles in deep holes etched in thick tem-plates to facilitate template-assisted growth of nanowires on (001) Si and InP substrates. In addition to a functional p-n junction on an (001) InP substrate, the first reported InP-InAs-InP axial heterostructures on an (001) substrate was demonstrated.

In order to reuse the substrate for multiple growth runs, growth of AlAs-GaAs nanowires for epitaxial lift-off was realized. Using selective gold electrodeposition, we demonstrated deposition of seed particles on a substrate after nanowire peel-off. AlAs-GaAs nanowires were subsequently regrown demonstrating the potential of this novel technique for substrate reuse.

Acknowledgment

It has been almost 8 joyful years from when I started my master thesis at the Division of Solid State Physics until today, when I am writing my PhD thesis. I have met so many amazing people. And I believe they have all helped me to make it to this point. My main supervisor Prof. Håkan Pettersson, I always appreciate the amount of time you dedicated to me and I am very thankful that you have let me do what I really loved. Your great support on the TA-VLS project was always motivating. Prof. Lars Samuelsson, what I owe you is far more than just to a project supervisor, your support both in research and personal life was incredible and I cannot express it in words. Last but not least, among my supervisors, Prof. Magnus Borgström, I have never felt that I am meeting my supervisor but it was like meeting an amazing friend. You have helped me a lot through my research from discussions on processing to my crazy questions on crystal growth. Thank you the incredible Magnus for always welcoming me when I showed up at your office door.

Prof. Jonas Johansson, thank you for all your help and nice discussions. Dr. Magnus Heurlin, thank you very much for all the efforts you put in my crazy projects, your impact was significant. I am grateful that we can still meet to support MFF – Heja Malmö. Dr. Enrique Barrigon, it was a pleasure to work with you and I truly appreciate all the collaborations and discussions we had, gracias. Anders and Mariusz, you taught me all practical and theoretical skills I needed in processing of nanostructures, you have had a huge impact on this work, thank you. Axel Persson, it was great to work with you, thank you for the nice discussions and the efforts you put into our collaboration. My old office mate and project partner, Dr. Jason Beech, thank you for all the nice discussions and help. Dr. Victor J. Gomez, it was always a pleasure to talk and work with you, thank you. Thank you Irene for the efforts you put in our work together. Prof. Anne L'Huillier, thank you for trusting me during my MSc. admission, you helped me to start this journey.

All of my colleagues in the Epi-group, Gaute, Alexander, Vilgaile, Xulu, Michel, Lukas, David, Yuwei, Aditya and Kristi, thank you for all the nice discussions we had. Thank you Mats-Eric for always caring about my projects. My officemate, Zhen, thank you for all great time we had. Dan, Claes, Christelle and Ivan, that was great to work as part of your teaching team, thank you. All my colleagues at FTF, Mohammad, Sebastian, Olof, Bekmorat, David G., Vishal, Anette, Sara T., Frida, Neimantas, Bao, Maryam, Oskar, Yang, Erik M., Elke, Therese, Johanna, Sudhakar, Antti, Hossein and Mercy, thank you for the great work environment you made.

LNL staff, Ivan, Håkan, George, Bengt, Sara M., Maria, Luke, Sungyoun, Alexander and David, thank you for all incredible efforts you put into keeping the LNL an amazing place to work. Peter and Dmitry, you guys were responsible for the main tools I used in the LNL. I cannot express how much I appreciate all your help in keeping those tools up and running, thank you very much. The administration office, Line, Mia, Anastasia, Helena and Abdul Rahman, thank you for all helps. My corridor mates, Veronique, Christian, Saeed, Anton, Marika, Cornelia, Lisa, Magdalena, Munzur, Belmin and Viola, thank you for all those five amazing years we had together. Hamid, Behruz, Mehran, Maryam, Milad and Niloufar thank you for always being supportive. Iman and Niousha, I appreciate all your help, thank you very much. My family in Lund, Sepideh, Omid and my little love Ava, words cannot explain my gratitude, thanks for being my family.

My mother and sister, we know how hard it was when I decided to start this adventure and besides all emotions, I know how hard it was for you to arrange all the financial support that I needed in order to make it. This would not have been possible without your support; you are my everything. Last but not least, my lovely wife Sara, you shined light into my life. You sacrificed all your needs for me when I was finishing this thesis. Your incredible support will always be appreciated, I love you.

Fall 2019

Reza

Popular science summary

In everyday life, we often use electrical and optical devices such as sources of green energy like a solar cell, or a multi-functional communication apparatus like a smartphone. Such devices consist of several components, mainly fabricated from semiconductors. Silicon (Si) is by far the most used semiconductor for fabrication of electronic devices. Consequently, the corresponding industrial process lines are fine-tuned with respect to the detailed material properties of Si. However, Si alone cannot satisfy all the requirements of todays' electronics. With increasing use of e.g. cloud computing and storage, and the emerging internet of things (IoT) paradigm, disruptive optical solutions are a necessity for fabricating high-speed, low-power devices. Silicon is the perfect candidate for fabricating waveguides, couplers and other passive components, but it does not efficiently emit light and therefore cannot satisfy all the requirements for current integrated optical circuits. III-V semiconductors, e.g. indium phosphide (InP) and gallium arsenide (GaAs), are made from group III and group V elements in the periodic table. These compound semiconductors have excellent optical properties such as a strong emission and absorption of light in a suitable wavelength region, which makes them ideal candidates for active on-chip optical components like lasers and photodetectors.

Due to a large lattice mismatch, it is still a challenge to integrate III-V components with Si wafers. Although there are several approaches to realize III-V/Si integration, there is still plenty of room for new approaches. Semiconductor nanowires (NWs) are thin needle-like nanoscale building blocks with the potential to realize integration of optical and electrical devices on Si wafers. The small diameter of the NWs accommodates the strain caused by mismatch when growing them on Si. The strain relaxation offered in a NW geometry can also be used to integrate different III-V materials in a single NW, so called heterostructures, to create new artificial materials with tailored optoelectronic properties. Gold (Au) particles are well-known catalyst seeds for growing NWs. There are several methods to deposit these Au particles on a substrate, but none of them offers selective deposition on patterned substrates. This drawback significantly increases how much Au is wasted during fabrication and in some cases even makes it impossible to fabricate the desired NW device.

This thesis deals with development and applications of a novel Au electrodeposition method for growth and processing of III-V NWs. Gold electrodeposition is a method to selectively deposit Au on any conductive area, and thereby significantly reduce the wastage of Au deposited on unwanted areas. We showed that we could reduce Au consumption by a factor of 650 compared to conventional deposition methods such as evaporation, while growing large NW patterns without any loss in quality. One issue related to growth of NWs is that they tend to grow inclined relative to the surface of standard commercially-available (001)-oriented substrates, which makes subsequent device processing troublesome. A recently reported method, developed by researchers at IBM, proposed to grow the NWs in deep holes etched in a template, effectively forcing them

to grow vertically from the substrate surface. Gold electrodeposition is the only way to selectively deposit Au seed particles at the bottom of such deep holes. We have recently demonstrated the first successful guided growth of InP NWs, InAs/InP NW heterostructures and InP-based diodes in deep holes etched and Au-seeded in SiO_x templates fabricated on (001) Si and InP.

Nanowires have been shown to be excellent candidates for fabrication of high-efficiency solar cells. Gallium arsenide NW solar cells with an efficiency of 16.8% have recently been reported. In order to make them competitive with Si solar cells on a broad market, they need to become cheaper to make. The main part of the final cost comes from the usage of expensive III-V substrates and the patterning steps required to prepare the substrates for growth. Using our developed method for selective electrodeposition of Au seed particles, in combination with a novel concept of introducing a sacrificial dissolvable segment at the bottom of the NWs, we have proposed a new approach to reuse the expensive substrates and avoid repeating some of the complex patterning steps. After growth of large ensembles of GaAs NWs with an embedded AlAs segment, the NWs are embedded in a polymer and removed from the substrate by a simple peel-off technique. The embedded NWs can be used for flexible solar cells or be transferred to a cheaper Si substrate. During peel-off, the NWs break at the AlAs segment. The remaining stubs of AlAs on the parent substrate can be dissolved in a chemical solution to make the NW surfaces flat. Subsequently, a new round of Au electrodeposition prepares the substrate for regrowth of AlAs segments followed by the GaAs NWs. In this way, the substrate can be reused many times and the cost of the final solar cell thereby reduced by up to 100 times.

List of Papers

- <u>R. J. Jam</u>, M. Heurlin, V. Jain, A. Kvennefors, M. Graczyk, I. Maximov, M. T. Borgström, H. Pettersson and L. Samuelson, "III-V Nanowire Synthesis by Use of Electrodeposited Gold Particles," *Nano Lett.*, vol. 15, p. 134–138, 2015. I led the project and performed all the lithography steps and gold deposition prior to the growth. I was actively involved in analyzing the results and wrote the paper with input from co-authors.
- II. <u>R. J. Jam</u>, A. R. Persson, E. Barrigón, M. Heurlin, I. Geijselaers, V. J. Gómez, O. Hultin, L. Samuelson, M. T. Borgström and H. Pettersson, "Template-assisted vapour-liquid-solid growth of InP nanowires on (001) InP and Si substrates", Nanoscale, 2019, DOI: 10.1039/c9nr08025b.
 I led the project and performed all the lithography steps, gold electrodeposition and growth of the NWs. I was actively involved in analyzing the results and wrote the paper with input from co-authors.
- III. <u>**R. J. Jam**</u>, Jason Beech, X. Zeng, Jonas Johansson, L. Samuelson, H. Pettersson and M. T. Borgström, "Embedded sacrificial AlAs segments in GaAs nanowires for substrate reuse", manuscript to be submitted.

I led the project and performed all the lithography steps, gold electrodeposition and growth of the NWs. I was actively involved in analyzing the results and wrote the manuscript with input from co-authors.

IV. V. J. Gómez, M. Graczyk, <u>R. J. Jam</u>, S. Lehmann, and I. Maximov, "Waferscale nanofabrication of sub-100 nm arrays by deep-UV displacement Talbot lithography", manuscript to be submitted.

I performed the gold electrodeposition and provided inputs on revised version of the manuscript.

Publications not included in this thesis:

V. V. Jain, M. Heurlin, E. Barrigón, L. Bosco, A. Nowzari, S. Shroff, V. Boix, M. Karimi, <u>R. J. Jam</u>, A. Berg, L. Samuelson, M. T. Borgström, F. Capasso and H. Pettersson, "InP/InAsP Nanowire-Based Spatially Separate Absorption and Multiplication Avalanche Photodetectors", *ACS Photonics* 2017, 4, 11, 2693-2698.

Abbreviations

AlAs	Aluminum arsenide
ALD	Atomic layer deposition
AsH ₃	Arsine
BCB	Benzocyclobutene
C_4F_8	Octafluorocyclobutane
CF ₄	Tetrafluoromethane
CHF ₃	Fluoroform
Cl	Chlorine
Cr	Chromium
Cu	Copper
DC	Direct current
DEZn	Diethylzinc
EBL	Electron beam lithography
EDS	Electron Dispersive X-ray spectroscopy
GaAs	Gallium arsenide
GaInP	Gallium indium phosphide
GaP	Gallium phosphide
H_2S	Hydrogen sulfide
HBr	Hydrogen bromide
HCl	Hydrochloric acid
HRTEM	High-resolution transmission electron microscopy
ICP-RIE	Inductively-coupled plasma reactive ion etching
InAs	Indium arsenide
InP	Indium phosphide
IPS	Intermediate polymer stamp
LOR	Lift-off resist
MOVPE	Metal-organic vapor phase epitaxy
N_2O	Nitrous oxide
NH ₃	Ammonia
NF ₃	Nitrogen trifluoride
NIL	Nanoimprint lithography
NWs	Nanowires
O ₂	Oxygen
PC	Pulse current
PECVD	Plasma enhanced chemical vapor deposition
PED	Pulse electrodeposition
PDMS	Polydimethylsiloxane

PL	Photoluminescence spectroscopy
PRC	Pulsed reverse current
RF	Radio frequency
RIE	Reactive ion etching
SEM	Scanning electron microscope
SF_6	Sulfur hexafluoride
Si	Silicon
SiH_4	Silane
SiN _x	Silicon nitride
SiO _x	Silicon suboxide
SiO ₂	Silicon dioxide
STEM	Scanning transmission electron microscopy
TASE	Template-assisted selective area epitaxy
TA-VLS	Template-assisted vapor-liquid-solid
TEM	Transmission electron microscope
Ti	Titanium
TMA1	Trimethylaluminum
TMGa	Trimethylgallium
TMIn	Trimethylindium
UV	Ultraviolet
UV-LED	Ultraviolet light-emitting diode
VLS	vapor-liquid-solid
VSS	vapor-solid-solid
WZ	Wurtzite
ZB	Zincblende

Introduction

Semiconductor nanowires, NWs, have great potential as an emerging technology in electronics and photonics^{1–8}. A leading technique used to grow semiconductor NWs is the vapor-liquid-solid (VLS) growth scheme in metal-organic vapor phase epitaxy (MOVPE).^{9–12} Gold is the most commonly employed metal for the catalyst in this growth process, and a wide range of different NW-based devices have been fabricated using Au as a catalyst.^{13–20}

There are several methods used to deposit Au, including aerosol techniques,^{21,22} colloid particles²³ and thermal evaporation²⁴ but they all have some significant drawbacks when applied to controlled array deposition, deposition in high-aspect-ratio structures and redeposition without excess lithography. When the technology for making highly efficient NW-based photovoltaics reaches its limits, it becomes increasingly important to reduce the fabrication cost by reusing the substrate. Substrate costs, lithography and Au consumption are all main components that determine the final cost of these solar cells. Cost reduction can be achieved by reusing the substrate in several growth runs, avoiding repeating lithography steps and using the optimum amount of Au in the process.

Electrodeposition has shown novel possibilities in nano-patterning with many important applications in semiconductor industry.^{25–30} Gold electrodeposition facilitates selective Au deposition in arrays and high aspect ratio features.^{28,31} Selective deposition helps when re-depositing Au and reusing the substrate without requiring additional lithography steps. The selectivity also significantly reduces Au consumption. Together, these lead to reduced material consumption and lower the cost of fabricating more complex structures. In addition, development of new deposition baths, particularly ones less toxic ones than cyanide baths, has made this process more environmentally friendly.

In 2008, Spurgeon et al. first demonstrated the use of Au electrodeposition for pillar growth.³² They showed the possibility of reusing the substrate up to four times with a 10% reduction in the growth yield. In their work, however, the diameter of the pillars was in the range of micrometers, and the report did not present a detailed study of the electrodeposition process or the crystal quality of the grown material.

This work presents a detailed study of electrodeposition, and investigates the potential of Au electrodeposition for semiconductor NW seed definition. Chapter 1 introduces the electrodeposition technique, giving a review of direct (DC) and pulse electrodeposition (PED), as well as discussing different Au electrodeposition baths. In Chapter 2, Au electrodeposition on III-V and silicon substrates is discussed in detail. Chapter 3 deals with NW synthesis by the VLS method from electrodeposited Au seed particles.

Several growth examples are presented, along with an extensive discussion of the effect of key deposition parameters on the properties of the deposited material.³³ This chapter also reports on template-assisted VLS (TA-VLS) growth of NWs on (001)-oriented InP and Si substrates.

Finally, Chapter 4 discusses the deployment of electrodeposition for reuse of substrates in fabrication of solar cells. Gallium arsenide NWs with a sacrificial aluminum arsenide (AlAs) segment are grown from electrodeposited Au seeds. After a mechanical NW peel-off process, the remaining AlAs stub is etched away. Subsequent deposition of Au seeds and regrowth of GaAs/AlAs NWs demonstrates a successful reuse of these InP substrates. Chapter 5 summarizes the work and provides an outlook for future development.

1 Gold electrodeposition

1.1. Introduction

Electrodeposition was investigated from the very earliest years of modern electrical science, soon after the invention of the Voltaic battery in the early 1800s.^{34,35} Together with copper (Cu), Au electrodeposition was the very first approach for surface metal coating. In addition to its color, Au has excellent material properties, such as corrosion resistance and high electrical conductivity, which make it suitable for a variety of applications. Gold electrodeposition was first applied to manufacture jewelry, but has expanded to other fields such as biomedicine and electronics.^{28,36,37} In the third quarter of 2019, over 66 tons of Au were used in the electronics industry,³⁸ and it is estimated that more than 50% of it was deposited by electrodeposition for wire bonding, electrical contacts, connectors and switches.

This chapter reviews the basics of electrodeposition, and describes the components required to carry it out. Different types of electrodeposition are discussed, as is the role of the various bath components and the effects of sample condition. A comparison is given between two common types of Au solutions, cyanide and sulfite baths.

1.2. Direct current electrodeposition

At its most general, electrodeposition is the process of coating the surface of a metal with another metal by applying an electric potential. An electric current flows between a cathode – the substrate – and an anode. The electrical circuit is completed by placing the anode and cathode in a conductive solution, which contains ions of the material to be deposited. The anode may be made of the same metal as the material to be deposited, but this is not the case in this work.

There are methods to perform deposition from an Au complex without applying an electrical potential, using so-called electroless deposition, but they are beyond the scope of this thesis.^{39–41}

Due to the applied current, the metal salt splits into ions. Positively charged metal ions, M^+ , will travel to the cathode, the negative pole of electrochemical cell, and are reduced solid metal, eq. (1.1).

$$M^{n+}(aq) + ne^{-} \to M(s) \tag{1.1}$$

The deposited mass, m, follows Faraday's laws of electrolysis:

$$m = \left(\frac{\text{QM}}{\text{FZ}}\right) \tag{1.2}$$

Where Q is the electrical charge in the bath, M is the molar mass, F is the Faraday number – the magnitude of electric charge per mole of electrons – and Z is the number of electrons transferred per ion. Considering that:

$$m = tAD$$
 (1.3) and
 $Q = iT$ (1.4)

Where Here t is the deposited thickness, A is the deposition area, D is the density of deposited material, i is the applied current and T is the duration of applied current. From this, (1.2) becomes:

$$t = \left(\frac{\text{iTM}}{\text{ADFZ}}\right) = \left(\frac{\text{JTM}}{\text{DFZ}}\right),\tag{1.5}$$

where J is the current density. Since all the other values are constants for a given material, the thickness of the deposited layer may be controlled by choosing a suitable current density and deposition time.

1.3. Choice of anode

The anode plays a crucial role in electrodeposition. Key anode parameters include the material from which it is made, and its dimensions. In some electrodeposition processes, the anode is made of the same material as the deposited coating. During such a process, the anode dissolves and replaces the material in the solution as it is used up in the coating. For example, the bath reactions in an alkaline Au cyanide bath with an Au anode are as follows:⁴²

Cathode:
$$[Au(CN)_2]^- + e^- \rightarrow Au + 2CN^-$$
 (1.6)

Anode:
$$Au + 2CN^- \rightarrow [Au(CN)_2]^- + e^-$$
 (1.7)

A well-known problem is that so-called 'sludge' can affect the smoothness of the deposition. Sludge is made up of particles from the anode material, which are detached due to non-uniform corrosion. In order to avoid any contribution of sludge in the process one should put the anode in an anode bag, usually made of polypropylene. Using high purity anodes can also decrease the amount of sludge.⁴³ The anode can also be made of another material, which eliminates sludge formation and results in a more stable process. In this case, however, the solution must be replenished to compensate for material consumption. The anode and cathode reactions for an alkaline cyanide bath are:⁴²

Cathode:
$$4[Au(CN)_2]^- + 4e^- \rightarrow 4Au + 8CN^-$$
 (1.8)

Anode:
$$H_2 0 \to 0_2 + 4e^- + 4H^+$$
 (1.9)

Full cell: $4[Au(CN)_2]^- + 2H_2O \rightarrow 4Au + O_2 + 4H^+ + 8CN^-$ (1.10)

For Au electrodeposition, platinized titanium (Ti) is the most commonly-used anode. Mesh anodes provide a better flow of solution, and hence a better uniformity of the deposit. The anode-to-cathode ratio, i.e. the ratio between the areas of the anode and cathode, is another important factor. At low anode-to-cathode ratio a higher rate of anode oxidation takes place, which is undesirable. It also affects solution depletion, which in turn affects deposition uniformity.

1.4. Electrodeposition baths

Cyanide is the most studied electrodeposition bath. However, because of health concerns, and the desired characteristics of the deposit, non-cyanide baths are used in many applications, especially in the electronics industry. In this section, we will review cyanide and non-cyanide Au solutions and discuss the advantages and disadvantages of each of them.

Cyanide baths

Cyanide baths are the most stable Au baths, but the presence of toxic HCN gas makes them less favored because of environmental and health concerns. Potassium and sodium salts are the most common form of cyanide complex. Potassium salts have a higher solubility. However, their very large negative reduction potential causes their cyanide baths to have a lower efficiency compared to others.⁴⁴

Acid cyanide baths

Acid cyanide baths typically have a pH less than 5. Their low pH makes acid baths compatible with photoresists, which is important for the electronics industry. However, one should note that even at low pH, cyanide may still affect the photoresist.⁴⁵

Acid cyanide baths have applications in industry for hard and bright Au deposition. Hard Au is the deposit alloyed with other metals to alter the deposit properties. Here it should be mentioned that low pH, or the use of additives such as hardeners or brighteners may increase the amount of carbon in the deposit. Research has shown that to decrease the amount of carbon, the bath temperature should be higher than 25 $^{\circ}C.^{42}$

Due to the lack of free cyanide in acid baths, Au cannot be oxidized from its metallic state, so anodes made of Au are not suitable, except when high current densities are used.

Alkaline cyanide baths

Working at a pH close to 10, alkaline solutions are not suitable for applications where resists and other polymers are involved. The Au deposits produced by this type of bath, which have a purity of more than 99.9%, are soft, and cannot be used where wear resistance is a concern. The excess of free cyanide in this bath does make it possible to use Au anodes, but may also affect the co-deposition of other metals because it alters their reduction potentials.

Neutral cyanide baths

With a pH between 5–8, and an absence of free cyanide, neutral baths are suitable for use with resists and polymers. At a pH greater than 5.5, the co-deposition of other metals is almost impeded, providing hard and soft Au deposits with a high purity.⁴² Additives can be used to alter the physical properties of the deposit, making it hard or bright. With proper parameter selection, there is no need for additives in a neutral cyanide bath, so additive concentration does not have to be controlled, and the bath will have more current efficiency.⁴⁶

Non-cyanide baths

Cyanide baths are good for Au electrodeposition in the electronics industry, because of the quality of the deposits and compatibility with photoresists, but there are health issues related to their high toxicity. Cyanide baths have a high stability, β =10³⁸, which means that in equilibrium, the electrolyte has a low tendency to oxidize at the anode or reduce at the cathode. Any alternative non-cyanide bath must exhibit a similarly high stability with a pH lower than 10.^{28,42,44} Sulfites and thiosulfates are the most common compounds for non-cyanide baths. They have low stability, β =10²⁷, compared to cyanide baths, but they show acceptable deposit properties and good compatibility with most photoresists. However, it is not possible to deposit hard Au from these baths.⁴⁷

Sulfite baths can produce deposits with good uniformity without effecting resists or polymers. They usually have a pH between 9–10 to retain compatibility for the microelectronics industry. Additives help to produce a more stable bath with a lower pH, but they can make waste disposal more difficult. For example, thallium acts as a grain refiner, but has toxicity issues of its own.⁴²

1.5. Pulse electrodeposition

Electrodeposition may be carried out by applying a DC current between the anode and the cathode, as discussed above, or by applying a modulated current amplitude in PED.^{48–52} Pulse current (PC) has an on time, T_{on} , followed by an off time, T_{off} . During T_{on} , the desired current density passes through the bath, while during T_{off} , no electrical current flows. Square-shaped pulses are often used because they provide a wide range of duty cycles and they do not need complicated pulse generators (see Figure 1.1). The duty cycle is defined as:

$$D = \frac{T_{on}}{T_{on} + T_{off}} \tag{1.11}$$

Another important pulse parameter is the average current density, J_A , defined as the current density at which the deposition rate of PED and DC electrodeposition is the same. J_A , is defined as:



Figure 1.1 Square pulse with basic parameters indicated

Where J_P is the peak current density. This means that for a pulse with a 20% duty cycle, the peak current should be 5 times higher than the DC level in order to have the same deposition rate as the DC electrodeposition.

Another important parameter for PED is the pulse frequency:

$$f = \frac{1}{\mathrm{T}} \tag{1.13}$$

Where T is pulse period, $T_{on} + T_{off}$.

Due to the nature of the electrodeposition process, the solution and sample will form a capacitor at their interface, a so called Helmholtz double layer. The simple electrical circuit analogy comprises a resistor in the form of the electrode-solution interface connected in parallel to this capacitor.⁵³ This capacitor first needs to be fully charged for the current to reach the desired level for deposition and then at the end of the pulse the capacitor needs to be fully discharged.^{54,55} Therefore, pulses with short rise and fall times, compared to the effective RC constant, lead to incomplete charging and discharging and non-optimal pulses. Figure 1.2 shows pulse shapes for a proper and improper choice of the on and off time, respectively.

The amount of ions reaching the cathode is governed by a diffusion layer close to the cathode which reduces the ion concentration during the on time. Figure 1.3 shows the simplified Nernst diffusion layer, indicating the ion concentration as a function of distance from the cathode. When the pulse is applied to the cell, the concentration of ions starts to decrease in the pulsating diffusion region (δ_p) until the pulse ends and the layer relaxes during the off time. The stationary diffusion layer (δ_s) extends to the bulk solution, and together with the pulsating diffusion layer form the entire diffusion layer. An applied pulse should be short to keep this diffusion layer as thin as possible but still long enough to fully charge the Helmholtz double layer.⁵⁵ One should not mix up the diffusion layer with the Helmholtz double layer, the former is usually several times wider than the latter and a matter of solution concentration rather than metal-to-solution interface.



Figure 1. 2:

a) Pulses with long rise/fall times lead to a fully charged double layer and proper current pulse formation. b)
 Pulses with too short rise/fall times lead to incomplete charging/discharging and non-optimal pulse formation.

In addition, since the pulse duration is typically short – in the millisecond range – one can apply a significantly higher current density to the bath compared to DC electrodeposition. A higher current density increases the density of nucleation points which gives a better uniformity in the final deposit. The on time controls the size of the grains, so a short pulse facilitates a uniform deposit with less porosity and a finer grain size. However, finer grain size is not always achievable since it also depends on the details of the bath chemistry.⁵⁶ One drawback of PED compared to DC electrodeposition is the higher cost of the current generator.



Distance from cathode

Figure 1. 3:

Schematics of Nernst diffusion layer. δN is the full length of the diffusion layer which is equal to the diffusion layer in DC electrodeposition.

Pulse reverse electrodeposition

Another type of PED utilizes a pulsed reverse current (PRC). There is no off time, instead two different current polarities are used: a cathodic current, J_C , and an anodic current, J_A , with durations of T_C and T_A , respectively. The average current density for during PRC is given by:

$$J_m = \frac{(J_C \times T_C) + (J_A \times T_A)}{T_C + T_A}$$
(1.14)

PRC has advantages over PC. During the anodic time, not only is the double layer discharged, but any uneven part of deposit is etched away, providing a better uniformity. High frequency PRC eliminates the need for additives, which improves the physical properties of the deposit. It is also helpful when depositing alloys with varying composition. There are several studies claiming that PRC can improve uniformity in deposition of high aspect ratio features.^{31,57,58}

1.6. Electrodeposition in the electronics industry

Electrodeposition offers several advantages for the electronics industry. It provides an economic way to deposit high quality material with the possibility to control various properties. Gold electrodeposition allows deposition of both soft and hard Au, which are required for high conductivity and hard wear resistance, respectively.^{25,59}

In through-mask deposition, material is electrodeposited using a mask, such as the resists used in electronics fabrication. The desired pattern is transferred to the substrate using lithography, but there are some challenges related to electrodeposition. Material being deposited needs to find its way to the smallest features, so the wetting properties of the resist play an important role. Usually, resists do not have good wetting properties, so organic additives are added to the solution to improve this.⁶⁰ In addition, resists are sensitive to alkaline solutions, with a pH around 10, which can make it difficult to deposit materials with the desired properties.

Another difficulty is related to the uniformity of the deposit. The deposit may be nonuniform around a particular feature, for example, if edges experience a higher deposition rate. This can be reduced by using PED. A deposition pattern may consist of features with different dimensions or different densities of features, but for uniform deposits it is recommended to design patterns with an even distribution of features. The overall bath geometry also has to be taken into account, with auxiliary electrodes often added to avoid current crowding at the edges.^{60,61}

Superfilling is a method to deposit void-free metal in high-aspect-ratio features.^{62–65} The difference between through-mask electrodeposition and superfilling is that in the latter there is always a seed layer deposited to ensure better nucleation. In superfilling, various additives are added to the solution to increase the deposition rate at the bottom of the holes in the mask, and lower it at the top and on the sidewalls, so that a uniform deposit is formed.

2 Gold electrodeposition for III-V nanowire seed definition

2.1. Introduction

Gold is the most studied seed particle material for VLS growth of NWs.^{10,66} Different methods of deposition have been used to define Au seed particles, all of which have their own drawbacks. While it is possible to generate small Au particles using aerosol technology, it is not possible to deposit them in an ordered array. Sputtering and evaporation can be used to deposit Au in lithographically-ordered arrays, but they cannot be used to deposit Au in high-aspect-ratio features, for example, when growing embedded NWs.

Metallization by evaporation and sputtering leads to significant amounts of material being wasted on the substrate and the sidewalls of the chamber. In thermal evaporation of thin films, around 1-2% of the evaporated material reaches the sample where the actual pattern usually covers only around 10-15% of substrate. This means that more than 98% of the material is wasted during the process. These methods also require low-pressure conditions, which makes them a low-throughput approach.

Gold electrodeposition, as a selective deposition approach, is a potential solution to all of these problems. It allows Au to be deposited in predefined places without any wasted material. Operation is under ambient conditions, making it a high-throughput approach. Electrodeposition will work with both organic resists, and inorganic masks (e.g. silicon dioxide (SiO_2) and silicon nitride (SiN_x)) if a suitable deposition bath is available. The use of inorganic masks facilitates reuse of the substrate, which is crucial for cost-efficient fabrication of many high-volume devices such as solar cells. There is no need for complicated process techniques to remove excess metal from the samples. Deposition in high-aspect-ratio structures for subsequent growth of complex embedded NW devices is also possible.

In this chapter, we will present our results on Au electrodeposition on Si and III-V substrates.

2.2. Gold baths

In this work, electrodeposition was carried out using a neutral cyanide bath, see section 1.4.1.3, with an Au purity of 99.9% (24K Pure Gold solution from Goldplating Inc.).⁶⁷ This bath has a pH of around 5.8, which is reasonably resist-compatible, although a certain degree of underplating does occur during the process as discussed in section 1.3. This solution also has a significantly lower price compared to electronics-compatible sulfite-based solutions. A platinized-titanium anode is suitable for this range of pH. About 950 ml of the solution is enough to coat a surface area of 8000 cm² to a thickness of 250 nm. For a reproducible process, the Au content in the solution should not decrease below 80% of the initial value without being replenished. The pH and bath temperature need to stay within 5% of their initial values. The pH can be adjusted by slowly adding 10% phosphoric acid to the bath.

The bath was heated to 35 °C to minimize carbon contamination, while retaining conditions as close as possible to ambient conditions. The purity of the deposited Au was examined by electron dispersive X-ray spectroscopy (EDS) and no traces of impurities were detected. The EDS in our transmission electron microscope (TEM) has a sensitivity of 1% so it cannot detect any signal if the concentration of the impurity is less than this.

2.3. Sample preparation

Plasma enhanced chemical vapor deposition (PECVD) and wet oxidation were used to provide the template on III-V substrates and Si, respectively. Nanoimprint lithography (NIL) and electron beam lithography (EBL) were used for patterning the resists, and the pattern was transferred to the underlying substrate with dry etching. In the following sections we briefly introduce each technique with its advantage and disadvantages.

Template deposition

PECVD was used to deposit the template $(SiN_x \text{ and } SiO_x)$ on III-V substrates while wet oxidation was used for growth of SiO₂ on Si substrates. Templates deposited with PECVD do not produce pure Si₃N₄ and SiO₂ so we use the sub-index x to donate the impurity of the deposit. Our technique usually results in a Si suboxide with Si deficiency.^{68,69} This could have an effect on the growth as it might lead to vacancies on the template surface, causing unwanted growth on the template.

Nanoimprint lithography

NIL was performed using an intermediate polymer stamp (IPS) in a simultaneous thermal UV process from Obducat AB.⁷⁰ NIL is a high resolution, high throughput method for large-scale pattern transfer.⁷¹ The pattern is created on a master nickel stamp, and then under specific conditions of pressure and temperature this stamp is pressed into a resist to generate the pattern on substrate. The technique has two main challenges. First, stamp and substrate separation can be difficult. Second, the stamp is affected by the process: stamps can break, and they may need to be cleaned or changed after a period of use, which adds cost to the process. To avoid this, it is possible to use the master stamp to fabricate an IPS replica. NIL is very sensitive, and it is hard to get the same results everywhere on the wafer. Even neighboring patterns may have different sizes due to imperfect processing or a defective stamp. This makes the subsequent electrodeposition and growth challenging. In this thesis the NIL resist is a negative epoxy resist of the TUx-xxx resist series in which UV light crosslinks the unpatterned parts, providing more stability for the unpatterned parts during the descum step where the remaining residues are removed by oxygen plasma etching.



Figure 2. 1:

Different steps in a NIL process. a) The sample is spin coated with LOR and NIL resists. b) The pattern is transferred to the sample by pressing the stamp into the NIL resist. c) Sample after development and d) SiN_x etching. e) Sample after removal of resists. f) Final metallization step by Au electrodeposition.

Electron beam lithography

EBL uses electrons to change the properties of the resist and transfer high-resolution patterns to the substrate. High-energy electrons interact with electron-sensitive resist molecules, both at the resist surface and deeper inside. In a positive resist, the exposed

part is then removed by dissolving in a developer; while in a negative resist, the exposed part is retained, and the unexposed part dissolves.

The resolution of EBL is limited by the proximity effect, caused by forward scattering and back scattering of electrons. The electron energy is a crucial parameter. For thick resists, a high energy is required, as otherwise the electrons are more strongly scattered, making the features broader, especially towards the base. Challenges for EBL are to provide a good beam quality over long exposure times. Also, beam drifting and differences in sample height can change the focus, and hence beam quality. The low throughput of the EBL system is the main disadvantage of this technique. The lack of any need for a mask makes it highly favorable in terms of pattern design.

Dry etching

Wet etching results in isotropic profile, which means the material will be etched in all directions, lateral and vertical. In device fabrication most of the time, an anisotropic profile is desired. Reactive ion etching, RIE, is a physical-chemical dry etching technique which helps to achieve an anisotropic etch profile. In this, process gasses under low pressure are turned into ions and the ions are accelerated towards a plate that holds the sample and is negatively charged by a radio frequency (RF) generator. In order to have a better control over the etching, inductively-coupled plasma reactive ion etching (ICP-RIE) is available. ICP-RIE has an extra RF coil which makes it possible to control the density of ions and have a better control over the etching profile. Dry etching takes place not only due to bombardment of the sample with high speed ions – the physical etch – but also by chemical reactions between substrate and the process gases. Ion scattering, and the chemical nature of etching affect the anisotropic profile, leading to some degree of isotropic etching. To compensate for this the etch sidewall could be passivated by a polymer to reduce lateral etching. The passivation could be by switching between etch and passivation gases (Bosch process) or by simultaneous introduction of all gases (steady state process).^{72–74} In the Si industry a cryogenic process is used to minimize the thickness of the passivation layer and increase selectivity towards the resist mask.^{75,76}



Figure 2. 2:

Etch profile of SiO₂ deposited on a) a Si substrate and b) an InP substrate. Evidently, the Si substrate is etched while the InP surface is left intact.

Dry etching of dielectrics

In this work, in order to etch the SiO_x and SiN_x , a fluorine-based recipe was used in which etching and passivation occur simultaneously. We used a CF₄ and CHF₃ gas mixture to etch SiN_x in a RIE system and a C₄F₈ and O₂ mix to etch SiO_x in an Oxford 100 ICP-RIE system. In deep etching of SiO_x, C₄F₈ acts as both etch and passivation gas while O₂ removes any redeposited carbon on the sample surface. The RF and ICP power were found to be important for deep etching of SiO₂. A high RF and ICP power is required to avoid RIE lag (low energy ions at the etch front) and to control sidewall passivation, respectively. These two parameters reduce the etch selectivity towards resist masks and a hard mask e.g. chromium (Cr) for deep etching is essential. As is evident from Figure 2.2, the recipe has a high selectivity to III-V substrate but a very low selectivity to Si. For Si substrates, over-etching should be avoided if it is necessary that the surface of the underlying substrate should remain unetched.

Dry etching of chromium:

In order to etch the hard Cr mask a standard chlorine based recipe is used. O_2 is very important in this process. If there is no O_2 in the reactor then the etch by-product is CrCl₂ which is volatile at around 1500 °C, way above the operation temperature of the ICP-RIE process. Adding O_2 to the process forms CrO₂Cl₂ as an etch by-product, which is volatile at room temperature. The trade-off is that O_2 is not desirable when a resist is used as the etch mask. Adding small amount of O_2 significantly improves the etch rate but above a 10% of gas mixture, O_2 degrades the etch rate.⁷⁷ The suitable value should be found for the equipment and the sample under test.



Figure 2. 3:

Etch profile of Cr a) using SiO_x as etch mask, b) e-beam resist as etch mask. the undercut does not occur with the e-beam resist. Etch masks are still in place.

A moderate RF power is required to etch Cr, we observed that the etch starts at RF power of 5 W and saturates at 20 W. The Cr etch rate is not significantly different in above mentioned window but the resist selectivity degrades significantly with increasing RF power. A RF power of 10 W was considered to be effective for a reasonable etch rate and selectivity. E-beam resist has a very low selectivity in the ICP-RIE system, which makes it hard to achieve deep etching with a resist mask. SiO_x or SiN_x have very

high selectivity in Cl-based recipe but in our experiments we always got an undercut in the profile which is not desirable, Figure 2.3. With the e-beam resist, the maximum etch depth was 85 nm but when a thicker template was needed, SiO_x was deposited by atomic layer deposition (ALD). For sub-100 nm holes we could not achieve etch rate more than 1 nm/min without use of ICP power unless we used a high RF power. However, high RF power reduces the selectivity more than ICP power. The story is different for trenches, where it is possible to etch sub-100 nm features without ICP power at a reasonable etch rate. It is well-known that trenches are etched faster due to higher transport of etch species and by-products into and out of the etch front. Although low pressure increases mask erosion, this process needs to be performed at low pressure: above 15 mTorr a keyhole profile was observed.

Substrate cleaning

Surface contamination can prevent material deposition or lead to poor adhesion, risking deposit peel off. Contamination can be extrinsic or intrinsic. Organic material or dust from the substrate are examples of extrinsic contamination, while native oxide is an example of intrinsic surface contamination.

Any resist residues from processing can lead to process failure. Silicon rapidly forms a native oxide when exposed to air, which makes it hard to electrodeposit high quality material on silicon. Organic residues can be removed by use of a solvent, but an aggressive cleaner, e.g. an acid, is required for the native oxide.

Acids such as HF will remove the native oxide, but the sample should be carefully rinsed afterward to remove any remaining acid, especially when using cyanide-based solutions. Electro-cleaning is another approach, where an electrochemical bath is used to clean the sample surface. Since the native oxide can form very rapidly, proper rinsing is important, and the choice of rinsing solution and environment is crucial. The rinsing solution should not facilitate oxide formation and it should not have any effect on bath properties.

Post deposition treatment is an important aspect, and simply rinsing in cold flowing water might not be enough to get rid of any residues from the solution. To achieve a cleaner surface, rinsing in hot water is suggested.

2.4. Deposition mechanism

Coherent material deposition may take place via a coalescence mechanism or a layer growth mechanism.⁷⁸ For Au electrodeposition, the deposition mechanism appears to be coalescence: the deposit starts by forming grains which then merge to form the final

deposit. The density and size of the grains are functions of applied current and deposition time respectively. To study the process mechanism, we interrupted a DC electrodeposition process on an n-type silicon substrate after a few seconds and inspected it by scanning electron microscope (SEM). As depicted in Figure 2.4 nuclei start by forming small grains which then merge to cover the whole surface. This confirms that a coalescence deposition mechanism occurs on the Si substrate.



Figure 2. 4:

The process begins by forming grains after a few secs (a). These then merge to form the final deposited Au seed particles (b). Scale bar is 500 nm.

2.5. Electrodeposition on Si substrates

Silicon has always been the material of most interest to industry. There are several previous investigations of electrodeposition on silicon.^{27,29,79} Huang et.al showed that grains will follow the surface property of the substrate for the first 50 nm, before becoming substrate-independent.⁸⁰ A clean and preferably intact surface is therefore needed for uniform and reproducible deposition. Here we present our results on electrodeposition on a Si substrate. Our substrate was highly n-doped, with a resistivity around 0.003 Ω cm.

Silicon forms a native oxide on its surface as soon as it is exposed to air, and this will affect electrodeposition. Since the process involves an interaction between the solution, the resist and the silicon surface, surface properties are important. There needs to be a high-quality hydrophilic surface with the proper wetting of profile sidewalls to permit the solution to pass through the holes in the mask and reach the Si surface.

Using EBL, a pattern of holes 100 nm in diameter and with a 1 μ m pitch was transferred to a Si wafer. Our experiments showed that the wetting property of the resist plays an important role, and the smaller the desired features, the harder it is for material to reach the surface. Without proper wetting, not all holes are filled with the same amount of Au. The native oxide was removed in a mixture of HF:H₂O (1:100) for 2 min followed by rinsing in 99% ethanol for 5 min. Ethanol does not oxidize the surface of silicon, and it dramatically improved the yield from electrodeposition. We believe that ethanol led to
an improvement in the wetting properties. The sample was then placed in water, for less than 3 s, to remove any ethanol. The sample was subsequently mounted to the rack without drying to keep surface wetting at its best. Figure 2.5 shows the results. All holes are filled with a deposit of the same thickness. Using a commercial electro-cleaning and surface activator to achieve better surface properties would be a good way to obtain more stable deposition.



Figure 2. 5:

a) Deposited Au particles with good uniformity on a Si substrate and b) with high yield as compared to the pattern.

2.6. Electrodeposition on III-V substrates

Direct electrodeposition

Resistivity is an important factor in electrodeposition, and a low resistivity leads to a better deposit properties.²⁹ Doping level and resistivity are in direct relation to each other because high doping results in high conductivity (low resistivity), so p^+ and n^+ InP, GaAs, GaP wafers were used in this work. A *n*-type InP wafer with a doping level of around 5×10^{18} cm^{-c}, corresponds to a resistivity around $5 \times 10^ \Omega$ cm. The difference in required current density for electrodeposition on different III-V substrates, regardless of feature size and doping type, comes from the difference in substrate conductivity.

Gold electrodeposition occurs selectively on conductive areas, hence there is no need to have a lift-off resist profile. However, TUx-xxx epoxy resists can leave some hard-to-remove residues on the substrate, making the surface less favorable for epitaxial processes. To avoid this, in the NIL process, we used a double layer resist profile consisting of a lift-off resist, LOR 0.7A, and a TUx-xxx NIL resist. The O₂ RIE process (descum) was continued until residues were removed from the pattern and the surface of the substrate was revealed.

A current density of 5.5 mA/cm² was applied to a *n*-type InP wafer with a doping level of around 5×10^{18} cm^{-c}. As is evident from Fig. 2.6a, the deposit forms a ring, with no

material deposited in the center of the opening. The reason is that sample was not conductive in the centers of the holes.



Figure 2. 6:

a) Deposition profile directly after descum and b) after HF cleaning. Resist mask is still in place. Scale bars are 500 nm.

The sample was placed in HF:H₂O (1:10) for 30 s and then the deposition process was repeated. Figure 2.6b shows the result: good deposits were observed across the whole sample. There are two possible explanations. Either the surface was oxidized during the descum process and this affected the center more than the edges, or the epi-ready oxide was removed at the edges but not at the center, where it blocked the deposition.

It is well known that a cyanide-based bath will gradually dissolve the resists and lead to under-plating. To avoid any under-plating a fast deposition process is required such that the sample is less exposed to the solution. We observed that for a PED process that lasts 250 s the under-plating is significant enough to affect the growth process. For process that involves a resist, in this case LOR 0.7, no under-plating was observed for a 10 s long direct plating.

Our measurements show that around 4.2 g of Au is needed to coat a 4-inch wafer to a thickness of 272 nm using a Pfeiffer Classic 500 evaporator, while the actual mass of Au on the final patterned substrate (with a pattern density of 6.25 holes/ μ m²) is around 6.4 mg. Thus electrodeposition potentially provides a reduction in Au consumption by a factor of 650. The required mass of 4.2 g of Au in the evaporator is independent of wafer size, so it is clear that for smaller wafers and samples there will be an even greater reduction in material consumption. Even if it is possible to recover half of the Au from the evaporation chamber sidewalls, a 300-fold reduction is still achieved, which represents a significant cost reduction for a high-volume fabrication line.

Electrodeposition on p-type substrates

Free electrons are required for electrodeposition. This is not a problem for n-type substrates, but in p-type substrates charge is carried by an excess of holes. For electrodeposition on p-type substrates, usually a metallic seed layer is deposited by another deposition method, but this is not applicable in NW technology. In our setup, we use a set of UV LEDs emitting at a wavelength of 405 nm to illuminate the substrate during processing and generate electrons (or rather, electron-hole pairs). Figure 2.7 below shows the result of the illumination on the electrodeposition. No deposition of Au is found in patterns without simultaneous light excitation.



Figure 2.7:

Deposition on p-type substrates, a) without IR illumination and b) with UV illumination. Scale bars are 100 nm.

Pulse electrodeposition on GaAs substrates

An n-type GaAs wafer was prepared to test the effect of PED on Au electrodeposition. As a reference, a peak current density of 5 mA/cm² was applied to the bath. To see the effect of the peak current on the properties of the deposit, current densities of 3 and 10 mA/cm² were also used. As expected, the deposit thickness increased with the peak current density, as is shown in Figure 2.8.



Figure 2.8:

Effect of peak current density on deposit thickness using PED, a) & b) 3mA/cm², c) & d) 5mA/cm² and e) & f) 10 mA/cm².

Theoretically, we expect to have the same thickness for PED as for DC deposition if the same average current density is applied.⁵² However, we observed a reduction in deposition rate from PED to DC deposition. To investigate this, we swept the pulse duty cycle. A longer duty cycle means more on time, and so a thicker deposit. Figure 2.9 depicts the effect of duty cycle on deposit thickness. Longer duty cycles do produce a thicker deposit, but the deposition rate reduces as the duty cycle increases. In these experiments, the pulse period was 50 ms with a peak current density of 5 mA/cm². For duty cycles of 10% and 20%, a 2-fold increase in deposition is expected, but the measured thickness increased by around 1.5 times. From 20% to 50% the factor is 1.25 and from 50% to 80% the factor is 1.2. The overall deposition time is constant, indicating a lower deposition rate as the duty cycle increases. The reduced deposition rate could be caused by charge build up at higher duty cycles, due to the longer on time. An on time less than 10 ms or an off time longer than 40 ms is optimum for this solution.



Figure 2.9:

Effect of duty cycle on deposition rate at a frequency of 20 Hz. The deposition rate decreases as the duty cycle increases.

To study the effect of the pulse frequency, samples were coated using frequencies of 100 Hz and 1000 Hz. The current density does not change with frequency, so the thickness is not expected to change. However, the deposition rate seems to be reduced by increasing the frequency at a constant duty cycle. Following the discussion about the diffusion layer in section 1.5, this means that either the on time is very long so the pulsating diffusion layer extends far towards the bulk or the off time is not sufficient to relax the pulsating diffusion layer. However, surface roughness seems to be improved by pulsing. Higher frequency with the same duty cycle means lower on and off time times, and hence a smaller grain size, which could lead to a smoother surface.

Figure 2.10 shows measurements of the potential between the anode and cathode. It is clear that for the n-type GaAs polished on both sides, the pulse has a good shape, while for n-type Si polished on one side only, the pulse shape is distorted: it takes time for the pulse to reach its peak value and then a longer time to relax to zero. This is related to the capacitance at the cathode-solution interface (see section 1.5 on Helmholtz double

layer), in the case of silicon the capacitance is high so it takes almost the entire on and off time to charge and discharge it. We tried to investigate this by examining other Si substrates. We oxidized a wafer and then removed the backside oxide to create a smoother back contact, which improved the pulse shape and increased the measured potential.



Figure 2. 10:

Actual pulse shape measured between the two electrodes. a) Double side-polished n-GaAs and b) one side polished Si.

A p-type GaAs wafer was prepared using NIL and light was shone on the sample during deposition. The deposition rate for p-type GaAs seems to be slower, despite the higher doping level than in the n-type samples. The measured voltage drop over the anode and cathode was lower, and the pulse shape showed the potential never reached its highest value. This explains the lower deposition rate. To obtain a thickness of 70 nm on a sample with an area equal to 1 cm^2 , a current density of 5 mA/cm^2 was applied for 4300 cycles, as shown in Figure 2.11.



Figure 2. 11:

Electrodeposited Au particles on p-type GaAs. a) The deposit filled the 70 nm thick SiNx template completely with b) a high deposition yield

2.7. Electrodeposition in high aspect ratio features

The challenge for through-hole plating is that material needs to penetrate the deep hole and reach the deposition site. A proper sidewall wetting is essential especially when dealing with sub 100 nm features. Etching polymer deposited on the sidewalls of the holes makes the wetting even worse. In case of Si it was easier to relax the wetting issue and achieve reasonable result. There are chemicals such as EKC 265 that can remove these polymers (Figure 2.12) and improve the cleaning of the native oxide. Rinsing in ethanol after HF dipping, improved the wetting properties even more. EKC 265 is incompatible with III-V substrates, making it hard to achieve a proper wetting and hence some degree of non-uniformity in deposit thickness was observed.



Figure 2. 12:

Effect of EKC 265 on removing the etch polymers. a) before and b) after using EKC 265. the polymers are significantely reduced.

We observed that a higher current density is required to deposit in deep holes compared to thicknesses below 150 nm. We also observed that the higher the aspect ratio, the higher the required current density. However, our results confirm that if the feature size is not in the sub-100 nm regime the aspect ratio dependency relaxes somewhat. This is because with larger feature sizes it is easier for the liquid to reach the deposition front. For sub-100 regime we have seen thickness variations among the deposited Au seeds between different holes and different samples, indicating that electrodeposition was not uniform and stable.



Figure 2. 13:

Electrodeposited Au on a 2-inch GaAs wafer, a) high yield and b) uniform deposit thickness.

2.8. Large scale electrodeposition

For deposition on larger, 2-inch wafers we used a deposition rack manufactured by Yamamoto-MS. When changing from a small bath to larger bath, it is important to take into account that the bath has a larger volume of material, 4 l, and the distance between the electrodes is different. As a result, a linear change in the current density is not sufficient to obtain the same deposit thickness. One needs to test and find the proper number of cycles as well. We observed that with the same current density as used for the small bath, we need almost half the number of cycles to achieve equal thicknesses. Deposition on 2-inch wafers was performed on samples prepared by NIL or Displacement Talbot lithography (Paper IV). Figure 2.13 shows our results for deposition on a 2-inch *p*-type GaAs wafer.

3 Nanowire synthesis

3.1. Introduction

There are many different techniques for nanowire synthesis, such as MOVPE, chemical beam epitaxy(CBE), and molecular beam epitaxy(MBE).⁸¹⁻⁸³ In this thesis we focus on NW growth using MOVPE. Possible growth modes include selective area (SA), VLS, and self-seeded growth; this thesis focuses on VLS.^{84,85} In this chapter we will discuss NW growth using deposited seed particles on different substrates, together with morphological and optical characterization. Subsequently, we present TA-VLS growth of NWs on (001) substrates by using electrodeposition to define seed particles in deeply etched high-aspect-ratio features.

3.2. Vapor-Liquid-Solid growth mode

Particle-assisted VLS growth is the original and most well-studied technique for NW synthesis.^{10,86} VLS uses a metal seed particle as catalyst to facilitate quasi one-dimensional NWs growth. The seed particle could be external, e.g. Au, or it could be self-seeding, as with In or Ga particles. External seed particles may be defined by various deposition methods, including electrodeposition.

Following deposition of the seed particles and removing any excess polymer, the substrate is moved to the growth reactor. The substrate is then heated to the desired growth temperature at which the effective pyrolysis of the metalorganic (MO) and halide precursors takes place. It has been asserted that the Au particle needs to be in the liquid state for growth to start, and so the growth temperature had to be above the eutectic temperature of the Au-substrate-vapor system. However, studies have confirmed that the seed particle does not need to be in the liquid state and the growth may be initiated in a vapor-solid-solid (VSS) mode provided that supersaturation at the seed particle is achieved.⁸⁷ However, the yield and growth rate are significantly affected under VSS as accommodation and precipitation in the liquid phase is considerably easier.⁸⁸

When the particle becomes supersaturated as a consequence of providing growth species that alloy with it, the material begins to precipitate from the seed particle and if an overpressure of vapor precursors is maintained the NW starts to grow underneath the seed particle. As the growth continues, the growth species do not only reach the crystal from the seed particle but will also react on the sidewalls as well. The growth is not purely unidirectional, but the growth rate on other facets is very low compared to the growth rate at the interface between the NW and its seed particle. Tapering due to lateral growth may occur, but can be impeded by introducing an etch gas such as hydrogen bromide (HBr) during growth.⁸⁹



Figure 3. 1:

a) 30° tilted SEM image of InP NWs grown using electrodeposited Au particles indicating a good surface morphology. b) Low-magnification SEM image showing excellent growth yield with respect to the defined mask.

3.3. Nanowire growth by use of electrodeposited Au particles

Growth on InP substrates

InP NWs were grown using of 33 nm Au particles deposited by DC electrodeposition. Figure 3.1 shows NWs grown using electrodeposited Au particles. In order to compare electrodeposition with conventional thermal evaporation, samples with the same parameters were prepared using evaporated metal particles. The NWs grown using electrodeposited Au particles have good surface morphology, comparable with the evaporated sample. The electrodeposited sample also has a high growth yield with respect to the growth pattern density

Single NW photoluminescence spectroscopy (PL) was used to study the optical characteristics of the NWs. Low temperature PL, Figure 3.2, shows that the NWs grown from electrodeposited Au seeds exhibit a spectral redshift compared to NWs grown from evaporated Au seeds.



Figure 3. 2:

PL spectra of InP NWs for different laser intensities. a) Spectra for samples prepared with evaporation (EV) and electrodeposition (EP) using a resist mask and b) spectra for samples grown with (red) and without (blue) a SiN_x template.

TEM inspection shows that both NW samples consist of a polytype mixture of both zincblende (ZB) and wurtzite (WZ) crystal phases. High-resolution transmission electron microscopy (HRTEM) confirmed that NWs grown from evaporated seed particles show less crystal mixing compared to those prepared by electrodeposition, see Figure 3.3.



Figure 3. 3:

HRTEM images of InP NWs grown from a) electrodeposited Au and b) evaporated Au particles. Photo courtesy of Magnus Heurlin.

The differences in observed crystal quality could be a result of different conditions during growth. The temperature could have drifted in different places of the MOVPE reactor, or the gas flow could be different if the substrates are placed in different positions. Both these explanations are plausible since we grow the NWs on the two substrates in the same run. The different polytypism could also stem from differences in the Au seed volume and wetting angle during growth. It is clear that the electrodeposited Au particles are more porous, which means the volume of the final particle, after annealing, is different than expected. It has also been shown previously that the wetting angle of the Au particle has a high impact on the crystal structure of the NWs.⁹⁰ It appears that postevaporation Au thickness is not a good measure for comparison, but Au volume should allow reliable comparison.

Growth on GaAs substrates

Both *n* and *p*-type GaAs wafers were coated with different thickness of SiN_x as a growth template. Gold with thickness ranging from 25–75 nm was deposited by either DC electrodeposition or PED. Figure 3.4 shows a micrograph of NWs grown using 25 nm thick Au on an n-type GaAs substrate, where high yield and good surface morphology are achieved. Low temperature single NW PL measurement confirmed a mixed crystal structure for these NWs.



Figure 3. 4:

Top view of electrodeposited Au particles on an n-type GaAs substrate using a 25 nm thick SiN_x template. b) Zoomed out SEM image shows a good yield of the deposited particles with respect to the deposition template. c)&d) 30° tilted SEM images of GaAs NWs gr from electrodeposited Au particles at two different magnifications.

Eliminating parasitic growth

An undercut resist profile is usually used in order to facilitate metal lift-off in a lithographic process. The undercut is formed by using a double layer resist profile comprising a bottom lift-off resist, LOR, and the desired resist for lithography on top. The LOR is then partially removed by use of a developer to make an undercut. During the subsequent lift-off process, the remover will more easily reach the resist through the undercut and remove the excess metal. When evaporation is used for seed definition, there is a risk that small particles attach to the undercut. This problem is even more severe for nitride templates, since scattered ions will affect the undercut region during the template etching. The attached Au particles can, depending on growth conditions, cause some parasitic growth. In order to eliminate this parasitic growth on GaAs substrate, Au electrodeposition was used. Figure 3.5 shows that the parasitic growth is completely eliminated by defining the Au seed particles with electrodeposition rather than by evaporation.



Figure 3. 5:

GaAs NWs grown from (a) evaporated seed particles and (b) electrodeposited seed particles. Parasitic growth is eliminated from the surface by using electrodeposited Au particles. Scale bars are 200 nm. Photo courtesy of Sebastian Lehmann.

Growth on GaP substrates

Hexagonal arrays of holes with 100 nm diameter and 1 μ m pitch was transferred to a highly doped *p*-type GaP wafer coated with 30 nm SiN_x. Gold particles were defined using DC electrodeposition. The Au thickness was measured to be 30 nm, as shown in Figures 3.6a and 3.6b. Figures 3.6c and 3.6d show successfully grown GaInP NWs on this substrate. The yield was not extremely high compared to the number of dots defined by the pattern, which we attribute to the fact that the growth parameters were tuned for evaporated Au particles, which have a different Au volume compared to electrodeposited particles. However, we cannot eliminate the possibility this was related to defects in the lithography.



Figure 3. 6:

a) Cross section and b) top view of array of Au particles electrodeposited on p-type GaP. c) GaInP NWs grown from the seed particles shown in part a) & b). d) NWs were grown with a very good yield. Images in c &d are by courtesy of Alexander Berg.

3.4. Template-assisted VLS growth on (001) substrates

Growth of NWs on (001) substrates, for example, on industrial standard Si(001), is a major challenge. Nanowires tend to grow in the <111>B direction so they usually develop kinks if grown on (001) substrates.^{86,91,92} Template-assisted selective area epitaxy (TASE) was used to grow InAs NWs on Si(001) substrates.^{93,94} In this approach a thick growth template was used to guide the NWs so they grow vertically on (001) substrates. Because of the selective area growth mode, the natural challenge with TASE is that if the template material is crystalline e.g. Si, then parasitic growth and antiphase boundaries are inevitable. This limits application of TASE on SOI wafers because the growth could happen simply on top of the device layers.^{95,96} Kum et.al used a two-step self-limited growth (TSSLG) mode to demonstrate defect free GaN NW based LEDs.⁹⁷ A self-seeding method was also used to grow nanospades in a template-assisted approach.⁹⁸ The main challenges with self-seeded mechanisms is that the seeding is random and the growth yield is very low and the growth need to be tuned.

Here we intend to develop an innovative technology to grow NWs on (001) substrates by VLS using a different template design. Using seed particles in NW growth can address the above-mentioned challenges, as the growth will only take place underneath the seed particle. Here we study template-assisted VLS(TA-VLS) using Au seeds to facilitate synthesis. With advances in alternative seed particles this approach could become compatible with mainstream Si platforms.^{99–101} Challenges include etching the template, defining the seed particles, maintaining template quality, and then growing high-quality material on (001) InP and Si to form heterostructures. Inside the template the growth species can reach the growth front from the seed particle. This limits the growth scheme to the seed particle properties inside the template.

Definition and quality of template

For the InP substrate some parasitic growth on the template surface was observed while on the Si substrate the surface of the template was clean. Any vacancy on the sample surface can act as a nucleation site and cause parasitic growth. To investigate this issue in more detail, SiO_x deposition was conducted at different temperatures. It was observed that the template deposited at 100 °C had less parasitic growth compared to the template deposited at 200 °C, as is shown in Figures 3.7a&b. No traces of any material other than Si and O₂ was detected in EDS inspection of the template material. Due to the limits of the EDS technique, we cannot eliminate the possibility that the templates had other elements than Si and O₂ or that there may be a Si or O₂ deficiency in the deposit.

As is evident in Figure 3.7c, deposition of a thin SiN_x layer on top of the template can significantly reduce parasitic growth, but at the expense of different growth conditions. As the SiO_2 on a Si substrate is grown by wet oxidation, the deposit is considered to be

stoichiometric and, as is shown in Figure 3.7d, results in no parasitic growth on the template surface.



Figure 3.7:

Effect of template quality on parasitic growth. a) SiO_x deposited at 100 °C, b) SiO_x deposited at 200 °C showed a higher amount of parasitic growth. c) A thin layer of SiN_x deposited at top of SiO_x improved the growth on the template surface. d) A stoichiometric SiO_2 template grown by wet oxidation on Si substrate completely eliminates surface growth.

Growth on InP substrates

In order to study TA-VLS growth of InP NWs in detail, a pattern with 100 nm openings on a 1 μ m pitch was transferred to a wafer with a 500 nm thick template. Growth on an (001) substrate requires that the seed particle fills the template diameter to ensure that the NW starts growing with a size as large as the opening and does not kink and crawl on the sidewalls during growth. Our experiment results were in agreement with theory predicting that the Au thickness needs to be around 66% of the radius of the initial cylinder to form a hemisphere with the same diameter as the opening. Following this, 65 nm thick Au particles were defined at the bottom of holes. In order to feed the seed particle with In and initiate the growth, the reactor was flushed with In while the PH₃ supply was shut down. A high amount of parasitic growth on the sample surface was observed. EDS inspection confirmed that the material is InP. The presence of In hemispheres on the template surface can be an indication of In-seeded growth on the template surface. The high degree of parasitic growth, and the fact that the NW growth rate at the template surface is higher than the growth rate inside the template, clogged some of the openings and led to non-uniform growth over the sample. Thus flushing was avoided and the TMIn molar fraction was increased (lower V-III ratio). Although parasitic growth was not completely eliminated, it was significantly reduced while the growth rate under the seed particle was increased.

Growth inside the template is time-dependent which is expected as the diffusion of growth species to the growth front is faster as the embedded NW reaches the template opening. Due to the natural tendency of NWs to grow towards the <111>B direction it

is expected that NWs kink when they come out of template but surprisingly they were mostly vertical with some degree of tapering, Figure 3.8a. It was shown that fast growth of NWs can lead to fewer defects in the crystal.¹⁰² This could explain the vertical growth of NWs when they emerge from the template. The growth rate outside the template is around 1 μ m/min and the crystal has far fewer stacking faults compared to the part grown inside the template.



Figure 3.8:

TA-VLS growth on InP substrates. a) 9 µm long InP NWs mostly grown vertical out of the template. b) TEM image of the NW part grown inside and c) outside of the template. The density of stacking faults reduced when grown out of the template. d) HRTEM of a NW close to the seed particle indicating a high quality ZB crystal. Inset shows the fft indicating a [001] crystal direction. Photos in part b-d are courtesy of Axel Persson

Despite the fact that unwanted growth on the template surface was inevitable, the grown NWs were shown to crystallize in the [001] direction, with a high density of stacking faults in the part grown inside the template, Figure 3.8 b-d. The difference in crystal quality in and outside of the template can be related to the fact that the NWs tend to grow towards the <111>B direction but the template is forcing them to grow vertically. Also, because the template profile has a positive slope, the Au particles diameter changes during growth which makes the growth mechanism unstable. As soon as NWs emerge from the template the Au particle becomes stable and govern the growth, resulting in a stable growth condition.

Growth with wider openings and a thicker template showed the same results, although the growth rate was higher due to faster diffusion in these holes. Figure 3.9 shows different SEM images of a substrate prepared for growth of 800 nm vertical NWs on a (001) InP substrate with an opening size of 250 nm, 500 nm pitch, and a template thickness of 1 μ m. As of previous results, the InP NWs have a [001] crystal orientation with a high density of stacking faults along its axis.



Figure 3. 9:

a) Gold deposition in deep via holes, b) Oxide template was removed to examine the deposition yield. c) InP NWs with 800 nm height were grown on a (001) InP substrate. d) Oxide was removed to check the morphology of the NWs.

Effect of dopants on TA-VLS

Introducing dopants to the reactor changes the growth scheme. First of all, the NWs do not grow vertically when they emerge from the template. This challenges the previous understanding that it might be rapid growth that led the NWs to grow vertically once out of the template.

In contrast to growth of free standing NWs, a high molar fraction of dopants did not degrade the growth yield inside the template. However, DEZn reduced the growth rate while H₂S increased it. To compensate for this the growth time for a *p*- segment was longer than *n* segment to obtain the desired length for each doping section. Using EBIC, a *p*-*n* junction was located inside the NWs, and the I-V curve showed a typical diode behavior with ideality factor of around η =2.6.

Growth of axial NW heterostructures using TA-VLS

Attempts to grow axial NW heterostuctures such as InP/InAs/InP failed when switching between desired group V precursors. Arsenic has a low vapor pressure and could react with any In particle trapped on the template surface instead of initiating NW growth inside the template. As an alternative, TMIn was shut down for a few seconds to deplete the Au particle from In. This led to growth of a pure InAs segment along the NWs, see Figure 3.10. The thickness of this segment was the same for a growth time of 20 and 60 s, which suggests that the amount of In in the particle reached the equilibrium value after about 20 s and the growth stopped. It would be interesting to study the dynamics of an Au particle at different states during this growth to investigate how it affects the growth conditions.

In conclusion, growth on InP substrates presents some challenges, including increasing the quality of the template and removing the Cr mask after deep etching. In addition, adhesion of oxide to the substrate is important: bad adhesion leads to delamination of

the oxide template during metal deposition. It would also be useful to study the doping profile of the NWs to help find their maximum possible doping level. This is an important concept as high doping of vertical NWs is still challenging and a requirement for electro-optical devices.



Figure 3. 10:

STEM/EDS image of a InP Nw with 100 nm embedded InAs segment. b-e colour code for In, P, As and Au, respectively. Photo courtesy of Axel Persson

Growth on Si substrates

 $1 \ \mu m$ thermal SiO₂ was grown on a n-type Si wafer in an oxidation furnace. A hexagonal pattern of holes with 100 nm diameters and a 1 μm pitch was transferred to the wafer. Si substrates are more challenging due to difficulties related to their native oxide. Also NWs need to have diameters of 100 nm, or less, to relax the lattice mismatch between Si and InP.



Figure 3. 11:

TA-VLS on 001 Si substrate. a) no parasitic growth on template surface. b) NW seems to be connected with substrate.

An important issue related to growth on Si is the formation of a native oxide on the Si surface. This oxide needs to be removed before growth to obtain a good interface between the grown material and the silicon substrate. Hydroflouric acid, HF, is a good candidate for this etching, but in this case it will etch the template as well, making the holes wider than desired. Studies have investigated how annealing at high temperature under AsH₃ overpressure can remove the native oxide.¹⁰³ Samples were annealed at 640–800 °C without and with an AsH₃ overpressure. Annealing at 800 °C under AsH₃ overpressure showed the best growth yield. As can be seen in Figure 3.11a, NWs are grown in all holes and there is no growth on the template surface. Room-temperature PL confirmed that the grown material is InP. HRTEM shows a ZB crystal structure with some diagonal twins. We do not yet understand why the crystal orientation is towards <111>; this could mean that the NWs do not have an epitaxial relation with the substrate. However, from Figure 3.11b it can be concluded that the growth indeed starts from the substrate surface.



Figure 3. 12:

Profile of a dry-etched hole after an additional 5 minute HF wet etch. This wet etch process makes a step shape at the interface between the Si substrate and the SiO₂ which can act as a second source for nucleation of both Au deposition and NW growth.

A reason for the imperfect growth could be that the etching of the thick template could not be controlled precisely enough to stop exactly on top of the surface, resulting in a slightly damaged Si surface. Figure 3.12 shows what an imperfect surface could look like. Profiles like this can lead to a failure of the deposition, or provide different nucleation points for metal deposition and NW growth. For high-aspect-ratio features, etching is often not uniform at the bottom of holes. Thus, holes will have slightly different profiles and sizes, which affects the Au thickness. The etch could also be stopped at some point above the surface, before etching the rest of the hole profile with wet chemicals to provide an intact surface, albeit at the expense of a wider opening size.

Growth of heterostructures on Si

We have also grown NW heterostructures in which half the NW is InP and the other half is InAs. These NWs were examined by TEM as shown in Figures 3.13a-b and were shown to be ZB crystallized in the [001] direction with some diagonal twins. Interestingly, as shown in Figure 3.13c, no trace of P was detected and thus the NWs were pure InAs. The reason for this was not investigated in detail but it could be that the incubation time (time to reach supersaturation) is longer than the time needed for growth of the InP segment.



Figure 3. 13:

TEM image of InAs NWs grown on (001) Si using TA-VLS. a) TEM overview of the entire NW b) The NW crystal is ZB with a crystal orientation towards [001]. Diagonal defects are evident. c) No trace of P was observed in EDS inspection although the sample was supposed to be InP/InAs.

4 Substrate reuse

4.1. Introduction

As the theoretical limits of fabricating devices with higher fundamental efficiencies are reached, the focus has shifted more towards fabricating devices that have a low overall cost. In III-V solar cell technology, the substrate makes the largest contribution to the cost, followed by lithography. Reusing substrates while avoiding additional lithography steps is thus one of the most interesting and important current research fields in device fabrication.^{32,104} One approach is to embed the grown material into a polymer and then peel them off from the growth substrate and transfer them to a cheaper substrate.

The peel-off process may be mechanical, in which the embedded NW-polymer composite is detached from the substrate by mechanical forces. This could lead to breaking of the polymer or the NWs, as well as leaving residues of the grown material behind, which would affect the next stage of growth. Epitaxial lift-off is another method, where a sacrificial layer of an easily removed material such as AlAs is grown first, and then the desired material is grown on top. The sacrificial layer is dissolved in an etchant, such as HF or HCl.^{105,106} In this chapter we discuss application and challenges of implementing AlAs in NWs for substrate reuse.



Figure 4. 1:

a) An InP substrate after peeling off the NWs. b) Electrodeposition of Au particles on sample shown in a). The remaining stubs are clearly visible. c) NWs grown on sample in b). d) Electrodeposition on a sample that was etched in HCI:H₂O (3:1). e) NWs grown on sample in d). f) Tilted cross-sectional SEM image of the

sample shown in e) after peel-off showing the remaining NWs in some part of the sample. Images in b)-d) are courtesy of Xulu Zheng.

Figure 4.1a shows a cross-section image of an InP substrate after mechanical lift-off. Figure 4.1b shows the reused substrate after electro-deposition of Au seed particles. Figure 4.1c shows the growth result, which does not have a satisfactory yield. This is because the substrate had some remaining stubs after the peel-off (Figure 4.1b). These stubs are longer than the template and cause the deposited Au to grow out of the SiN_x template and migrate across the surface during annealing and growth. Figure 4.1d shows an Au-deposited sample where the remaining stubs were etched with a mixture of HCl:H₂O (3:2) for 20 s. The challenge here was to control the etch profile. The etch rate is very high, even for a very dilute mixture. This wet etch gives an isotropic profile and thus an etching of the material underneath the SiN_x template. The deposited Au then faces different barriers during growth, affecting the final growth property, as shown in Figure 4.1e. In addition, there are polymer residues after peel-off, and NWs remaining behind, seen in Figure 4.1.f. These affect both Au deposition and NW growth

The remaining polymer residues can be dissolved in a remover. The challenge is to etch away the remaining NW stubs without losing the template fidelity. As can be seen from Figure 4.1a, the stubs have oblique profiles, making them more challenging to etch uniformly at all points. It has been shown that during annealing, Au particles can dissolve some InP from the substrate.⁷¹ It could therefore be useful to first deposit a thin layer of Au, and anneal it to erode the remaining stubs, and then deposit more Au to perform the growth. However, a precise control over the Au thickness and amount of group III during annealing is hard to achieve.

An alternative is to use a sacrificial layer for epitaxial lift-off. This section presents our results on application of AlAs NWs and Au electrodeposition for substrate reuse. The intention here is to grow a short AlAs NW and then grow the desired NW device on top. However, as has been shown before, growth of GaAs on AlAs is challenging and NWs will kink.¹⁰⁷

4.2. Growth of GaAs/AlAs nanowires on GaAs substrates

In contrast to bulk epitaxial lift-off, a SiNx template is required to avoid excess lithographical steps. A 100 nm thick template was deposited on a p-type GaAs substrate using PECVD and was patterned by NIL. Seed particles were defined by use of thermal evaporation and lift-off. The properties of the SiNx template are very important. Adhesion to the substrate is a major problem, the template may delaminate at high temperature or be washed off under flowing water. This limits the growth temperature to an upper limit of 450 °C which is way below the reported growth for AlAs NWs in MOVPE and MBE.^{107,108} When the Au seed particles are Al rich (little to no Ga is accommodated) then they are in a solid phase and the process is VSS. This makes it challenging to start growth from the GaAs substrate, and in order to form a liquid particle, growth of a short GaAs nucleation stub was inevitable, see Figure 4.2a. AlAs NWs were grown on top of this stub, Figure 4.2.b.



Figure 4. 2:

a) Growth profile after growing an GaAs stub. A difference in particle shape and hence wetting angle is evident. b) AIAs NWs are grown with high yield as compare to the growth template.

Although the growth had a reasonable yield, the growth rate was very low compared to e.g. GaAs. The growth rate is around 5 nm/min while GaAs is typically grown at a rate of about 150 nm/min. This difference could be either related to very low migration length of the Al species compared to Ga, or it could be that this process takes place in a VSS mode which has a significantly lower growth rate than VLS.⁸⁸ AlAs seems to grow quite fast in the lateral direction for a 600 nm long AlAs section, the measured diameter was 300 nm. HBr was employed to reduce lateral growth. HBr not only etch the lateral growth but it also improves particle stability, leading to a better growth yield.⁸⁹ As expected, HBr reduced the growth rate from 10 nm/min to 5 nm/min but sig-nificantly helped to increase the growth yield. However, a high amount of HBr can significantly reduce the growth rate of AlAs.

The central part of the sample had slightly lower growth yield compared to the edges. Changing the V/III ratio did not significantly improve this, it seems that a high V/III ratio was required which was not possible as the low TMAl molar frac-tion results in almost no growth of AlAs. As the yield was not significantly improved by different V/III ratio, we conclude that either the bad growth is due to a VSS growth mode, or more probably, the wetting properties of the particles are different as it is evident that some of the particles have a different wetting angle, Figure 4.2 a. The NIL stamp has some defects which can result in some deviation in profile of neighboring holes. This causes some of the openings to have different growth conditions.

Finally, 2.5 μ m long GaAs NWs were grown on top of an AlAs segment. The final growth yield is determined by the AlAs growth: where the AlAs has a high growth yield the final AlAs-Gas NWs were also grown with a high yield, as shown in Figure 4.3a.

Figure 4.3b is a cross section of the sample in which a contrast between the 500 nm AlAs segment and the 2.5 μ m GaAs NW is evident. A shell growth of AlGaAs was observed in this step, to avoid it the HBr molar fraction was slightly increased to etch the shell away.



Figure 4. 3:

a) 30 degree tilted view of AlAs-GaAs axial NWs with high yield as compare to the patterned template. b) Cross-sectional SEM micrograph of the same sample showing the uniform growth. The contrast between AlAs and GaAs is clear.

4.3. Nanowire peel-off

In order to investigate epitaxial lift-off, the NWs were embedded in a double layer of LOR 0.7 and Benzocyclobutene (BCB). LOR acts as a sacrificial layer while BCB holds the GaAs NWs. A developer can remove the LOR and leave the NWs embedded in the upper BCB layer.



Figure 4. 4:

Status of a sample embedded in a double layer of BCB and LOR 0.7A. After 24 hours of development in MF319 just a few mm was removed.

This approach has an issue with that the penetration of developer into the pattern takes a long time and the developer can affect the template. Figure 4.4 shows a SEM image of a section close to the edge were the MF319 etched away the LOR. After 24 hours only a few millimeters were removed. A higher quality template could improve the low fidelity and provide a way to investigate epitaxial lift-off.

In this study the NWs were embedded in Polydimethylsiloxane (PDMS) and removed from the substrate. The sample was subsequently rinsed in Dynasolve 220 to remove any remaining PDMS residues.

4.4. Etching of sacrificial AlAs segment

The major challenge here is how to remove the AlAs. The most well-known etchant for this process is HF, but it would etch the SiN_x template very rapidly so an alternative HCl:H₂O (1:1) solution was chosen. HCl etches GaAs (001) very slowly but it can etch the (111) substrates to some extent, to minimize this effect it was diluted with water.



Figure 4. 5:

a) SEM image of a sample after peeling off the NWs. Residues from the AlAs are evident. b) After etching the AlAs away, all NW surfaces are flat and the template pattern is preserved.

Figure 4.5a shows the sample status after peel-off while Figure 4.5b shows the sample after etching the remaining AlAs segments away. Conventional peeling without a sacrificial layer will leave an oblique profile in the peeled NWs as well, which makes it challenging for subsequent device fabrication. An advantage of a sacrificial layer is that the bottom of the NWs after removing the AlAs is flat which is ideal for device fabrication.

4.5. Gold electrodeposition

In order to deposit the seed particles for regrowth electrodeposition was employed. The main challenge is the bottom GaAs stub; it does not completely fill the template opening so Au particles can position themselves anywhere on its surface. This makes regrowth challenging, and particles positioned on sidewalls could cause parasitic growth. Figure 4.6 shows the status after Au deposition; it is clear that due to deposition on sidewalls and underplating, the deposited Au is not enough to form a uniform deposit. As the Au solution cannot dissolve the SiN_x template, a reason for under-plating is that the template was not properly adhering to the underlying substrate and when it was placed in the solution it simply delaminated, allowing Au to find its way and underplate. A more careful observation in Figure 4.2a reveals that at the edge of the pattern the template is slightly delaminated, this could strengthen the above mentioned hypothesis. We strongly believe that this happened during the annealing at 450 °C.



Figure 4. 6: Sample after Au deposition for substrate reuse; underplating is evident.

4.6. Regrowth of AlAs/GaAs nanowires

As expected from the seed definition, the regrown material does not have a high yield. This is for two main reasons. First of all, the template was delaminated at some points. This happens because of the under-plating of Au and its movement underneath the template during annealing. This could be significantly improved by using a high quality template.



Figure 4.7:

30 degree tilted SEM images of a) GaAs NWs and b) AlAs-GaAs NWs after regrowth. Pattern is preserved but the growth yield is affected by the under-plating and parasitic growth.

The second reason for low yield is deposition on the sidewalls of the GaAs stubs which can cause parasitic growth on the sample surface. However, the pattern is still preserved. Figure 4.7a shows the results of GaAs NW regrowth, the pattern is indeed preserved but parasitic growth on the surface is evident. Figure 4.7b shows the regrowth results for AlAs-GaAs NWs; as expected the yield is slightly lower as compared to the GaAs growth but still the pattern is preserved.

In conclusion the approach looks promising but there are challenges involved. The great achievement is that the template pattern fidelity is preserved after peeling off the NWs. Provided that the template has a good adhesion to the substrate, this means that the pattern fidelity could be preserved, making unlimited reuse a possibility. Since the template has a significant effect on all process steps the template quality and adhesion to the substrate both need to be significantly improved. The GaAs nucleation stub at the bottom seems to be a challenge. In the first instance, deposition on the sidewalls needs to be avoided. This could be improved by increasing the thickness of the Au particle so it fills the pattern completely. In order to fill the entire pattern, it might be possible to deposit the GaAs stub by electrodeposition.^{109,110} On the other hand, the role of particles during growth depends on the amount of Ga it contains, so during regrowth the nucleation needs to start with Ga to grow the NWs. Maybe a better template allows for a higher growth temperature in which the growth mode is VLS, eliminating the need for GaAs nucleation stub.

5 Conclusions and Outlook

In this work, we have developed a selective Au electrodeposition technique for definition of seed particles in patterns for semiconductor NWs synthesis. Both organic resists and inorganic SiN_x and SiO_x masks were used to define patterns on silicon and III-V substrates. Gold was selectively electrodeposited from a neutral cyanide-based Au bath. In the case of p-type substrates, Au was electrodeposited by simultaneously illuminating the sample with light. Using electrodeposition, we demonstrated a 650 times reduction in material consumption with a significant improvement in throughput, as compare to thermal evaporation.

Pulse electrodeposition was implemented and shown to reduce the porosity of the deposited Au seeds. A smoother Au coating was achieved by using pulse electrodeposition rather than DC electrodeposition. The measured pulse shape, however, indicated issues with incomplete charging and discharging of the capacitor related to the electrode-solution interface, the main reason being a significant back contact resistance between the substrate and cathode holder.

III-V semiconductor NWs were grown on both n- and p-type substrates with characteristics and yield comparable with those grown from seed particles defined by thermal evaporation. From careful TEM inspection and PL measurements we cannot find any evidence of differences in quality of NWs grown from seeds deposited by electrodeposition and evaporation.

Moreover, we have developed TA-VLS to enable growth of InP NWs on (001) Si and InP substrates using a template. After electrodepositing Au seed particles at the substrate surface, vertical NWs were grown in the holes by MOVPE. The grown NWs have a [001] crystal direction with stacking faults. Using TA-VLS we also demonstrated guided growth of InP NW p-n junctions and axial InP-InAs-InP NW heterostructures. Further investigations of sample preparation are needed to improve the uniformity and reproducibility of the Au electrodeposition. In particular cleaning of the holes in the template is crucial for a successful growth of guided NWs with a high crystal quality and yield.

Finally, an AlAs sacrificial layer together with Au electrodeposition was employed for substrate reuse in fabrication of NW photovoltaics. After peeling off GaAs NWs comprising an AlAs segment from the substrate, the AlAs was dissolved in an etchant and Au seeds were deposited on short GaAs nucleation stubs on GaAs substrates. The subsequent regrowth of GaAs/AlAs NWs showed that the pattern is preserved. The yield, however, was not high mainly due to challenges involving the template properties. The results of this work suggest that Au electrodeposition could be developed into a key resource in semiconductor NW technology with significant advantages compared to alternative deposition techniques such as evaporation.

For the next step TA-VLS on (001) Si needs to be investigated further. The clean surface of the template provides an easy process for contacting the NWs without requiring excessive lithography. From a lithography perspective, deep etching processes need to be investigated in more detail to achieve a more anisotropic profile as the lithography step seems to effect the crystal quality of the grown material. Etching must stop at the top of the substrate surface to keep it intact. A more detailed study on TA-VLS growth mechanism is also required to reduce the growth on the template surface for InP substrates. Different strategies need to be studied to improve surface wetting and uniform material diffusion into the deep trenches during electrodeposition.

Regarding the substrate reuse a higher quality template with reduced stress for better adhesion to the substrate is a necessity. A more detailed study of the growth mechanism would be helpful to understand what causes some of the AlAs NWs to crawl on the surface. The usage of PDMS is not cleanroom-compatible wherefore an epitaxial liftoff method could be a solution. Embedding NWs in a cleanroom compatible polymer and optimizing the solution penetration for faster penetration of etch species to the etch front is recommended in this case.

6 References

- 1 X. Duan, Y. Huang, R. Agarwal and C. M. Lieber, *Nature*, 2003, **421**, 241–245.
- 2 L. Samuelson, M. T. Björk, K. Deppert, M. Larsson, B. J. Ohlsson, N. Panev, A. I. Persson, N. Sköld, C. Thelander and L. R. Wallenberg, *Physica E: Low-dimensional Systems and Nanostructures*, 2004, 21, 560–567.
- 3 R. Yan, D. Gargas and P. Yang, *Nature Photon*, 2009, **3**, 569–576.
- 4 P. Yang, R. Yan and M. Fardy, *Nano Lett.*, 2010, **10**, 1529–1536.
- 5 M. T. Borgström, J. Wallentin, M. Heurlin, S. Fält, P. Wickert, J. Leene, M. H. Magnusson, K. Deppert and L. Samuelson, *IEEE Journal of Selected Topics in Quantum Electronics*, 2011, **17**, 1050–1061.
- B. Piccione, C.-H. Cho, L. K. van Vugt and R. Agarwal, *Nature Nanotech*, 2012, 7, 640–645.
- 7 X. Guo, Y. Ying and L. Tong, Acc. Chem. Res., 2014, 47, 656–666.
- 8 E. Barrigón, M. Heurlin, Z. Bi, B. Monemar and L. Samuelson, *Chem. Rev.*, 2019, **119**, 9170–9220.
- 9 L. E. Jensen, M. T. Björk, S. Jeppesen, A. I. Persson, B. J. Ohlsson and L. Samuelson, *Nano Lett.*, 2004, **4**, 1961–1964.
- 10 V. G. Dubrovskii, G. E. Cirlin, N. V. Sibirev, F. Jabeen, J. C. Harmand and P. Werner, *Nano Lett.*, 2011, **11**, 1247–1253.
- 11 M. Heurlin, M. H. Magnusson, D. Lindgren, M. Ek, L. R. Wallenberg, K. Deppert and L. Samuelson, *Nature*, 2012, **492**, 90–94.
- 12 K. Tomioka, J. Motohisa, S. Hara and T. Fukui, *Nano Lett.*, 2008, 8, 3475–3480.
- 13 L. J. Lauhon, M. S. Gudiksen, D. Wang and C. M. Lieber, *Nature*, 2002, **420**, 57–61.
- 14 Y. Cui, Z. Zhong, D. Wang, W. U. Wang and C. M. Lieber, *Nano Lett.*, 2003, 3, 149– 152.
- 15 L.-E. Wernersson, E. Lind, L. Samuelson, T. Löwgren and J. Ohlsson, *Jpn. J. Appl. Phys.*, 2007, **46**, 2629.
- 16 E. D. Minot, F. Kelkensberg, M. van Kouwen, J. A. van Dam, L. P. Kouwenhoven, V. Zwiller, M. T. Borgström, O. Wunnicke, M. A. Verheijen and E. P. A. M. Bakkers, *Nano Lett.*, 2007, 7, 367–371.
- 17 M. D. Kelzenberg, D. B. Turner-Evans, B. M. Kayes, M. A. Filler, M. C. Putnam, N. S. Lewis and H. A. Atwater, *Nano Lett.*, 2008, 8, 710–714.
- 18 C. P. T. Svensson, T. M\aartensson, J. Träg\aardh, C. Larsson, M. Rask, D. Hessman, L. Samuelson and J. Ohlsson, *Nanotechnology*, 2008, 19, 305201.
- 19 M. H. M. van Weert, N. Akopian, U. Perinetti, M. P. van Kouwen, R. E. Algra, M. A. Verheijen, E. P. A. M. Bakkers, L. P. Kouwenhoven and V. Zwiller, *Nano Lett.*, 2009, 9, 1989–1993.

- 20 J. Wallentin, N. Anttu, D. Asoli, M. Huffman, I. Åberg, M. H. Magnusson, G. Siefer, P. Fuss-Kailuweit, F. Dimroth, B. Witzigmann, H. Q. Xu, L. Samuelson, K. Deppert and M. T. Borgström, *Science*, 2013, **339**, 1057–1060.
- 21 P. Caroff, M. E. Messing, B. M. Borg, K. A. Dick, K. Deppert and L.-E. Wernersson, *Nanotechnology*, 2009, 20, 495606.
- 22 M. E. Messing, K. Hillerich, J. Bolinsson, K. Storm, J. Johansson, K. A. Dick and K. Deppert, *Nano Res.*, 2010, 3, 506–519.
- 23 A. I. Hochbaum, R. Fan, R. He and P. Yang, *Nano Lett.*, 2005, 5, 457–460.
- 24 K. Hiruma, K. Haraguchi, M. Yazawa, Y. Madokoro and T. Katsuyama, *Nanotechnology*, 2006, **17**, S369–S375.
- 25 I. R. Christie and B. P. Cameron, *Gold Bull*, 1994, **27**, 12–20.
- 26 E. J. Menke, M. A. Thompson, C. Xiang, L. C. Yang and R. M. Penner, *Nature Mater*, 2006, 5, 914–919.
- 27 L. D. V. Llona, H. V. Jansen and M. C. Elwenspoek, J. Micromech. Microeng., 2006, 16, S1–S6.
- 28 T. A. Green, *Gold Bull*, 2007, **40**, 105–114.
- 29 T. Fujita, S. Nakamichi, S. Ioku, K. Maenaka and Y. Takayama, *Sensors and Actuators A: Physical*, 2007, **135**, 50–57.
- 30 M. A. Imtaar, A. Yadav, A. Epping, M. Becherer, B. Fabel, J. Rezgani, G. Csaba, G. H. Bernstein, G. Scarpa, W. Porod and P. Lugli, *IEEE Transactions on Nanotechnology*, 2013, 12, 547–552.
- 31 P. Dixit and J. Miao, J. Electrochem. Soc., 2006, 153, G552–G559.
- 32 J. M. Spurgeon, K. E. Plass, B. M. Kayes, B. S. Brunschwig, H. A. Atwater and N. S. Lewis, *Appl. Phys. Lett.*, 2008, **93**, 032112.
- 33 R. Jafari Jam, M. Heurlin, V. Jain, A. Kvennefors, M. Graczyk, I. Maximov, M. T. Borgström, H. Pettersson and L. Samuelson, *Nano Lett.*, 2015, 15, 134–138.
- 34 Electrodeposition of Alloys 1st Edition, https://www.elsevier.com/books/electrodeposition-of-alloys/brenner/978-1-4831-9808-8, (accessed November 9, 2019).
- 35 L. B. Hunt, *Gold Bull*, 1973, **6**, 16–27.
- 36 Y. Okinaka and M. Hoshino, *Gold Bull*, 1998, **31**, 3–13.
- 37 R. Steegmueller, C. Wagner, T. Fleckenstein and A. Schuessler, Gold Coating of Nitinol Devices for Medical Applications, https://www.scientific.net/MSF.394-395.161, (accessed November 9, 2019).
- 38 Gold Demand Trends Q3 2019, https://www.gold.org/goldhub/research/gold-demand-trends/gold-demand-trends-q3-2019, (accessed November 9, 2019).
- 39 J. Hu, W. Li, J. Chen, X. Zhang and X. Zhao, Surface and Coatings Technology, 2008, 202, 2922–2926.
- 40 Y. Okinaka and M. Kato, in *Modern Electroplating*, John Wiley & Sons, Ltd, 2011, pp. 483–498.
- 41 J. C. Whelan, B. I. Karawdeniya, Y. M. N. D. Y. Bandara, B. D. Velleco, C. M. Masterson and J. R. Dwyer, *ACS Appl. Mater. Interfaces*, 2014, **6**, 10952–10957.
- 42 P. A. Kohl, in *Modern Electroplating*, John Wiley & Sons, Ltd, 2011, pp. 115–130.
- 43 1960.

- 44 A. M. Sullivan and P. A. Kohl, J. Electrochem. Soc., 1997, 144, 1686–1690.
- 45 H. Honma and K. Hagiwara, J. Electrochem. Soc., 1995, 142, 81-87.
- 46 H. A. Reinheimer, J. Electrochem. Soc., 1974, 121, 490–500.
- S. Dimitrijević, M. Rajčić-Vujasinović and V. Trujić, *Int. J. Electrochem. Sci.*, 2013, 8, 27.
- 48 Ch. J. Raub and A. Knödler, *Gold Bull*, 1977, **10**, 38–44.
- 49 G. Devaraj, S. Guruviah and S. K. Seshadri, *Materials Chemistry and Physics*, 1990, 25, 439–461.
- 50 P. C. Baldwin, *Metal Finishing*, 1995, 93, 24.
- 51 S. Mohan and V. Raj, *Transactions of the IMF*, 2005, 83, 72–76.
- 52 M. S. Chandrasekar and M. Pushpavanam, *Electrochimica Acta*, 2008, **53**, 3313–3322.
- 53 J. J. McMullen and N. Hackerman, J. Electrochem. Soc., 1959, 106, 341.
- 54 J. Cl. Puippe and N. Ibl, *J Appl Electrochem*, 1980, **10**, 775–784.
- 55 N. Ibl, Surface Technology, 1980, 10, 81–104.
- 56 T. Pearson and J. K. Dennis, *Transactions of the IMF*, 1991, **69**, 75–79.
- 57 J.-M. Lee and A. C. West, J. Electrochem. Soc., 2005, 152, C645–C651.
- 58 P. Dixit, J. Miao and R. Preisser, *Electrochem. Solid-State Lett.*, 2006, 9, G305–G308.
- 59 Y. Okinaka, *Gold Bull*, 2000, **33**, 117–127.
- 60 G. Zangari, in *Modern Electroplating*, John Wiley & Sons, Ltd, 2011, pp. 617–636.
- 61 J. O. Dukovic, *IBM Journal of Research and Development*, 1990, **34**, 693–705.
- 62 D. Josell, B. Baker, C. Witt, D. Wheeler and T. P. Moffat, *J. Electrochem. Soc.*, 2002, **149**, C637–C641.
- 63 T. P. Moffat and D. Josell, J. Electrochem. Soc., 2012, 159, D208–D216.
- 64 D. Josell and T. P. Moffat, J. Electrochem. Soc., 2013, 160, D3009–D3014.
- 65 D. Josell and T. P. Moffat, J. Electrochem. Soc., 2013, 160, D3035–D3039.
- 66 R. S. Wagner and W. C. Ellis, *Appl. Phys. Lett.*, 1964, 4, 89–90.
- 67 24k Pure Gold Solutions, https://www.goldplating.com/collections/24k-pure-gold-solutions, (accessed August 25, 2019).
- 68 P. G. Pai, S. S. Chao, Y. Takagi and G. Lucovsky, *Journal of Vacuum Science & Technology A*, 1986, **4**, 689–694.
- 69 S. S. Chao, J. E. Tyler, Y. Takagi, P. G. Pai, G. Lucovsky, S. Y. Lin, C. K. Wong and M. J. Mantini, *Journal of Vacuum Science & Technology A*, 1986, **4**, 1574–1579.
- 70 T. Eriksson, S. Yamada, P. Venkatesh Krishnan, S. Ramasamy and B. Heidari, *Microelectronic Engineering*, 2011, 88, 293–299.
- 71 G. Otnes, M. Heurlin, M. Graczyk, J. Wallentin, D. Jacobsson, A. Berg, I. Maximov and M. T. Borgström, *Nano Res.*, 2016, 9, 2852–2861.
- 72 F. Laerme, A. Schilp, K. Funk and M. Offenberg, in *Technical Digest. IEEE International MEMS 99 Conference. Twelfth IEEE International Conference on Micro Electro Mechanical Systems (Cat. No.99CH36291)*, 1999, pp. 211–216.
- 73 R. Abdolvand and F. Ayazi, Sensors and Actuators A: Physical, 2008, 144, 109–116.
- 74 M. D. Henry, phd, California Institute of Technology, 2010.

- 75 Ü. Sökmen, A. Stranz, S. Fündling, H.-H. Wehmann, V. Bandalo, A. Bora, M. Tornow, A. Waag and E. Peiner, *J. Micromech. Microeng.*, 2009, **19**, 105005.
- 76 K. A. Addae-Mensah, S. Retterer, S. R. Opalenik, D. Thomas, N. V. Lavrik and J. P. Wikswo, *Journal of Microelectromechanical Systems*, 2010, 19, 64–74.
- H. Nakata, K. Nishioka and H. Abe, *Journal of Vacuum Science and Technology*, 1980, 17, 1351–1357.
- 78 M. Paunovic, M. Schlesinger and D. D. Snyder, in *Modern Electroplating*, John Wiley & Sons, Ltd, 2011, pp. 1–32.
- 79 G. Oskam, J. G. Long, A. Natarajan and P. C. Searson, J. Phys. D: Appl. Phys., 1998, 31, 1927–1949.
- Q. Huang, H. Deligianni and L. T. Romankiw, J. Electrochem. Soc., 2006, 153, C332– C336.
- 81 L. Samuelson, C. Thelander, M. T. Björk, M. Borgström, K. Deppert, K. A. Dick, A. E. Hansen, T. Mårtensson, N. Panev, A. I. Persson, W. Seifert, N. Sköld, M. W. Larsson and L. R. Wallenberg, *Physica E: Low-dimensional Systems and Nanostructures*, 2004, 25, 313–318.
- 82 A. I. Persson, M. W. Larsson, S. Stenström, B. J. Ohlsson, L. Samuelson and L. R. Wallenberg, *Nature Mater*, 2004, 3, 677–681.
- 83 M. D. Brubaker, P. T. Blanchard, J. B. Schlager, A. W. Sanders, A. M. Herrero, A. Roshko, S. M. Duff, T. E. Harvey, V. M. Bright, N. A. Sanford and K. A. Bertness, *Journal of Elec Materi*, 2013, 42, 868–874.
- 84 K. Tomioka, K. Ikejiri, T. Tanaka, J. Motohisa, S. Hara, K. Hiruma and T. Fukui, *Journal of Materials Research*, 2011, 26, 2127–2141.
- 85 R. L. Woo, L. Gao, N. Goel, M. K. Hudait, K. L. Wang, S. Kodambaka and R. F. Hicks, *Nano Lett.*, 2009, 9, 2207–2211.
- 86 K. Hiruma, M. Yazawa, K. Haraguchi, K. Ogawa, T. Katsuyama, M. Koguchi and H. Kakibayashi, *Journal of Applied Physics*, 1993, 74, 3162–3171.
- 87 B. A. Wacaser, K. A. Dick, J. Johansson, M. T. Borgström, K. Deppert and L. Samuelson, *Advanced Materials*, 2009, 21, 153–165.
- 88 F. M. Ross, C.-Y. Wen, S. Kodambaka, B. A. Wacaser, M. C. Reuter and E. A. Stach, *Philosophical Magazine*, 2010, **90**, 4769–4778.
- 89 A. Berg, K. Mergenthaler, M. Ek, M.-E. Pistol, L. R. Wallenberg and M. T. Borgström, *Nanotechnology*, 2014, 25, 505601.
- 90 F. Glas, J.-C. Harmand and G. Patriarche, *Phys. Rev. Lett.*, 2007, **99**, 146101.
- 91 U. Krishnamachari, M. Borgstrom, B. J. Ohlsson, N. Panev, L. Samuelson, W. Seifert, M. W. Larsson and L. R. Wallenberg, *Appl. Phys. Lett.*, 2004, 85, 2077–2079.
- 92 V. Schmidt, S. Senz and U. Gösele, Nano Lett., 2005, 5, 931–935.
- 93 M. Borg, H. Schmid, K. E. Moselund, G. Signorello, L. Gignac, J. Bruley, C. Breslin, P. Das Kanungo, P. Werner and H. Riel, *Nano Lett.*, 2014, 14, 1914–1920.
- 94 M. Borg, H. Schmid, K. E. Moselund, D. Cutaia and H. Riel, *Journal of Applied Physics*, 2015, **117**, 144303.
- 95 H. Kim, A. C. Farrell, P. Senanayake, W.-J. Lee and D. L. Huffaker, *Nano Lett.*, 2016, 16, 1833–1839.

- 96 I. Giuntoni, L. Geelhaar, J. Bruns and H. Riechert, Opt. Express, OE, 2016, 24, 18417– 18427.
- 97 H. Kum, H.-K. Seong, W. Lim, D. Chun, Y. Kim, Y. Park and G. Yoo, Scientific Reports, 2017, 7, 40893.
- 98 L. Güniat, S. Martí-Sánchez, O. Garcia, M. Boscardin, D. Vindice, N. Tappy, M. Friedl, W. Kim, M. Zamani, L. Francaviglia, A. Balgarkashi, J.-B. Leran, J. Arbiol and A. Fontcuberta i Morral, ACS Nano, 2019, 13, 5833–5840.
- 99 T. I. Kamins, R. Stanley Williams, D. P. Basile, T. Hesjedal and J. S. Harris, *Journal of Applied Physics*, 2000, **89**, 1008–1016.
- 100 V. T. Renard, M. Jublot, P. Gergaud, P. Cherns, D. Rouchon, A. Chabli and V. Jousseaume, *Nature Nanotechnology*, 2009, 4, 654–657.
- 101 K. A. Dick and P. Caroff, Nanoscale, 2014, 6, 3006–3021.
- 102 H. J. Joyce, Q. Gao, H. H. Tan, C. Jagadish, Y. Kim, M. A. Fickenscher, S. Perera, T. B. Hoang, L. M. Smith, H. E. Jackson, J. M. Yarrison-Rice, X. Zhang and J. Zou, *Nano Lett.*, 2009, 9, 695–701.
- 103 K. Tomioka, Y. Kobayashi, J. Motohisa, S. Hara and T. Fukui, *Nanotechnology*, 2009, 20, 145302.
- 104 E. Yablonovitch, T. Gmitter, J. P. Harbison and R. Bhat, *Appl. Phys. Lett.*, 1987, **51**, 2222–2224.
- 105 P. Demeester, I. Pollentier, P. D. Dobbelaere, C. Brys and P. V. Daele, Semicond. Sci. Technol., 1993, 8, 1124.
- 106 C.-W. Cheng, K.-T. Shiu, N. Li, S.-J. Han, L. Shi and D. K. Sadana, *Nature Communications*, 2013, 4, 1577.
- 107 K. A. Dick, S. Kodambaka, M. C. Reuter, K. Deppert, L. Samuelson, W. Seifert, L. R. Wallenberg and F. M. Ross, *Nano Lett.*, 2007, 7, 1817–1822.
- 108 A. Li, D. Ercolani, L. Lugani, L. Nasi, F. Rossi, G. Salviati, F. Beltram and L. Sorba, *Crystal Growth & Design*, 2011, 11, 4053–4058.
- 109 I. Villegas and J. L. Stickney, J. Electrochem. Soc., 1992, 139, 686–694.
- 110 M.-C. Yang, U. Landau and J. C. Angus, J. Electrochem. Soc., 1992, 139, 3480–3488.


Advanced patterning and processing for III-V nanowire device fabrication

This thesis describes applications of gold (Au) electrodeposition in the synthesis of III-V semiconductor nanowires (NWs) and related devices. We have shown that Au electrodeposition can successfully replace conventional methods of seed deposition such as thermal evaporation.

The top left image in the figure shows an SEM micrograph of Au seed particles defined on a p-type InP substrate using a resist as the deposition mask. The Au particles are coloured artificially. The top right image is a cross-sectional SEM micrograph of a GaAs-AlAs-GaAs NW heterostructure which is used to facilitate substrate reuse. One of the NWs is artificially coloured to highlight the different materials along the NW.

The bottom left image is a cross-sectional SEM micrograph of a NW p-n junction grown on a p-type (001) InP substrate using TA-VLS. This approach is promising for integration of NWs with the (001) substrates used in mainstream industry. The p and n sections are artificially coloured in different shades of red. The bottom right image is a cross-sectional SEM micrograph of an InP-InAs-InP NW heterostructure grown by TA-VLS on a p-type (001) InP substrate. The estimated position for the InAs segment, as confirmed by STEM, is artificially coloured in blue.



Department of Physics Faculty of Engineering LUND UNIVERSITY

